

# Temi di Discussione

(Working Papers)

A public guarantee of a minimum return to defined contribution pension scheme members

by Giuseppe Grande and Ignazio Visco







# Temi di discussione

(Working papers)

A public guarantee of a minimum return to defined contribution pension scheme members

by Giuseppe Grande and Ignazio Visco

Number 762 - June 2010

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

*Editorial Board:* Alfonso Rosolia, Marcello Pericoli, Ugo Albertazzi, Daniela Marconi, Andrea Neri, Giulio Nicoletti, Paolo Pinotti, Marzia Romanelli, Enrico Sette, Fabrizio Venditti. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

### A PUBLIC GUARANTEE OF A MINIMUM RETURN TO DEFINED CONTRIBUTION PENSION SCHEME MEMBERS

by Giuseppe Grande\* and Ignazio Visco\*

### Abstract

The recent financial crisis has clearly demonstrated the exposure of defined contribution (DC) pension scheme members to extreme financial market risks. This paper argues that the government might offer DC plan members a minimum return guarantee, funded by risk-based premia. Option pricing formulas show that the guarantee could be quite expensive, but public provision could reduce the costs borne by workers. Such an arrangement would be sustainable for the government and would give workers an acceptable benefit/contribution ratio in worst-case scenarios, while still allowing them to reap the advantages of occupational or individual funded pension schemes.

JEL Classification: D10, G23, H55, J14.

**Keywords**: defined contribution pension schemes, financial market tail risks, return guarantees, exchange option, Modigliani's Treasury CFDB swap.

### CONTENTS

1.	Introduction	5
2.	Mixed defined benefit pension schemes	6
3.	Defined contribution pension schemes with forms of financial protection	8
4.	A public minimum return guarantee for defined contribution schemes	9
5.	Long-run growth in GDP and share prices: a comparison	12
6.	The advantages for the worker of a government-guaranteed minimum return	14
7.	The risks of a guarantee fund	16
8.	An assessment of the premium payable by the pension scheme members	20
9.	Why should the guarantee be provided by the government?	27
10.	Conclusions	29
App	bendix A: The data used in the estimates of guarantee fund's maximum outlays	30
App	bendix B: The method of estimating the option value	31
App	bendix C: The value of the option with a lump-sum investment	32
Refe	erences	33

<sup>\*</sup> Bank of Italy. The corresponding author is Giuseppe Grande: <u>giuseppe.grande@bancaditalia.it</u> (address: Via Nazionale, 91 – 00184 Roma – Italy; tel.: +390647922643; fax: +390647923720).

### **1.** Introduction<sup>1</sup>

In a typical defined contribution (DC) pension scheme the investment risk is borne entirely by the plan member. This basic characteristic of DC schemes has long been acknowledged and underscored both by researchers on social security systems and by experts of economic policy authorities.<sup>2</sup> Involving as it does the transfer of financial risks from firms or from the public sector to workers, this feature should be a concern for government authorities. In fact, sample surveys reveal a low level of financial education among consumers, even in countries where pension funds have been in existence the longest.<sup>3</sup> It is also likely that exposure to investment risk goes some way to explain the low levels of membership and contributions to these schemes in a number of countries.<sup>4</sup>

The management of financial investment risks is a rather complex task, even for specialized intermediaries. This is because the prices of financial assets are subject to what are known as tail risks, in other words to the tendency to generate extreme losses with a greater frequency than what "chance" would be expected to generate in normal conditions. A clear demonstration of how vulnerable any pension scheme based on the accumulation of financial assets is to stock exchange downturns was seen in the recent financial crisis. For the United States, for instance, there is evidence that in 2008 many 401(k) participants nearing retirement (aged 56-65) had very high exposures to equities and suffered large reductions in their account balances (on the order of 25 per cent).<sup>5</sup> Somewhat more reassuring is the evidence provided by Gustman, Steinmeier and Tabatabai (2009) using data from the Health and Retirement Study. Rather low stock market holdings and substantial reliance on defined benefit (DB) plans as primary pension are likely to have muted the impact of the financial crisis on the wealth of US households in which at least one member is approaching retirement. Nevertheless, there is little dispute that future generations of retirees will rely on DC plans much more than current generations. Moreover, as Coile and Levine (2009) stress, severe financial turbulences are typically associated with weak labour markets and if older job seekers have difficulty finding work, they may retire earlier than expected (rather than delaying retirement), resulting in lower pension income.

<sup>&</sup>lt;sup>1</sup> We are greatly indebted to Hideki Konishi, Filippo Taddei and Juan Yermo for very helpful discussions. We are grateful for useful comments from Henk J. Brouwer, Riccardo Cesari, Nicolas A. Cuche-Curti, Antonio Di Cesare, Lorenzo Forni, Maura Francese, Daniele Franco, Andrea Gerali, Laurence J. Kotlikoff, Giuseppe Marotta, Sandro Momigliano, Joana Pereira, Tommaso Perez, Antonio Rossetti, Paolo Sestito, Mauricio Soto, Daniele Terlizzese, Anita Tuladhar, Roberto Violi and seminar participants at the Bank of Italy, the Center for Advanced Research in Finance of the University of Tokyo, the CeRP 10th Anniversary Conference (Collegio Carlo Alberto, Moncalieri) and the conference on "Risk Sharing in Defined Contribution Pension Schemes" held at the University of Exeter on 7-8 January 2010. The views expressed herein are those of the authors and do not necessarily reflect those of the Bank of Italy. The authors are solely responsible for any errors.

<sup>&</sup>lt;sup>2</sup> See, for example, Diamond (1977), Bodie (1990), Fornero (1999), Blake (2000), Campbell and Feldstein (2001), Orszag and Stiglitz (2001), the Group of Ten (2005) and the Pensions Commission (2005).

<sup>&</sup>lt;sup>3</sup> See OECD (2005), and Lusardi and Mitchell (2007). For an analysis of some of the evidence in Italy, see Cesari, Grande and Panetta (2008).

<sup>&</sup>lt;sup>4</sup> For a discussion of the low level of contributions to pension funds and the high degree of exposure to financial risks of members of DC pension funds, see Visco (2009).

<sup>&</sup>lt;sup>5</sup> See VanDerhei (2009). Changes in account balances reflect not only investment returns but also contributions, which increase the rate of accumulation. For cross-country evidence on the impact of the crisis on pension funds, see Antolín and Stewart (2009).

The objective of avoiding or reducing workers' exposure to financial risks is one of the reasons for the development of mixed pension schemes, such as "hybrid" DB funds or DC funds with forms of protection of returns. Moreover, in some countries, public authorities provide rate of return guarantees. In the wake of a dramatic financial crisis, it is vital to consider whether the forms of protection from fluctuations in the prices of financial assets now available to DC plan members can actually provide an effective shield against extreme investment risks. This calls for greater use of intergenerational risk sharing (Bütler, 2009) and, as with the unresolved problem of aggregate longevity risk,<sup>6</sup> possible forms of public intervention (Visco, 2009).

This paper assesses the hypothesis of the government providing DC plan members with the guarantee of a minimum return. The guarantee would be compulsory, but each plan member could choose among different levels of minimum return depending on the maximum loss that he or she would be willing to bear. DC plan members should pay risk-based premia to a guarantee fund, which would be able to prefund its activity.

Two aspects should be highlighted to place things in proper perspective. First, our proposal is not only fully consistent with the private provision of occupational or individual funded pension schemes, but is designed to strengthen the DC pillar of retirement systems, as it would allow governments not to leave workers alone to deal with extreme financial market risks. Second, the advantage of introducing a public minimum return guarantee depends critically on the design of the overall retirement arrangements; it is less relevant for countries where public pensions represent the bulk of retirement income.

We start by briefly examining the main characteristics of mixed DB pension schemes (Section 2) and mixed DC schemes (Section 3). Section 4 presents the guarantee mechanism. Section 5 shows some statistics illustrating the long-term relationship between GDP growth and trends in financial markets. Then the paper analyzes the benefits and risks of the proposed guarantee scheme (Sections 6 and 7, respectively). Section 8 addresses the issue of how to determine the market value of the guarantee, and Section 9 summarizes the reasons why government intervention might be necessary. Section 10 concludes.

### 2. Mixed defined benefit pension schemes

Several DB pension schemes aim to shelter members from investment risks, but establish some relationship between contributions paid in and benefits (a typical characteristic of DC funds) in order to favour fund solvency.

Three notable examples of these mixed pension schemes (also known as "hybrid") are: (1) cash balance plans, which in the past decade have expanded significantly in the United States; (2) collective pension schemes in the Netherlands; and (3) the centralized DB pension fund proposed in the first half of this decade by Franco Modigliani and his co-authors.

CASH BALANCE PLANS. To all effects and purposes, these are DB schemes, in which the investment risk is borne entirely by the pension fund sponsor (typically the employer). Like all other defined-benefit plans cash balance plans are guaranteed,

<sup>&</sup>lt;sup>6</sup> See Group of Ten (2005), Visco (2007), and Antolín and Blommestein (2007).

within certain limits, by the Pension Benefit Guaranty Corporation. What sets them apart from the others is the fact that the amount of pension benefits does not depend on parameters such as pay or length of service but on two other variables: the contributions paid in and a predetermined capitalization rate (which can be fixed or variable). In fact, in a cash balance plan, the contributions paid in by the worker (directly, or by the employer on his behalf) are capitalized at a previously set rate, which in turn can be indexed to another variable (for example, the yield on 1-month US Treasury bonds). The benefit the worker receives at retirement will therefore depend on the total amount of contributions and this capitalization rate. Members' contributions to cash balance plans are invested on financial markets by the employer; it is therefore the employer who is responsible for all the entire investment risk and return.<sup>7</sup>

*COLLECTIVE PENSION SCHEMES IN THE NETHERLANDS.* Dutch collective pension schemes have a mechanism of intergenerational risk-sharing. Through a combination of increases in contributions and cuts in indexation of pension benefits, funding deficits are shared among currently employed people, pensioners and future generations. The benefits of this and other mechanisms of intergenerational solidarity that characterize Dutch collective pension schemes have to be assessed against the costs of restrictions of individual choices and possible labor market distortions.<sup>8</sup>

THE MODEL OF MODIGLIANI ET AL. The model for a pension system proposed by Modigliani and his co-authors arose primarily from the concern that workers should not be left alone to cope with investment risks.<sup>9</sup> According to the proposal, the fulcrum of the retirement system should be a centralized and fully funded scheme, that is financed by mandatory contributions and that offers defined benefits (contributory funded defined benefit scheme, CFDB scheme). In this system, the workers (and their employers) pay in contributions to the fund, which credits them to individual accounts. These payments are managed on a collective basis (by government or by private intermediaries delegated for this purpose); once a year (or in any event periodically), the pension fund effects a swap with the Treasury, whereby the earnings accrued in the year through the management of the portfolio are swapped with those that would have been generated if the return on the portfolio's management had been equal to a previously determined amount (5 percent, in Modigliani and Muralidhar's proposal of 2004). This swap assures all workers that the contributions to the CFDB fund are capitalized at a fixed rate and that the investment risk will be borne entirely by government.<sup>10</sup>

<sup>&</sup>lt;sup>7</sup> The cash balance accounts differ from typical DB funds in other ways too, such as the procedure for paying out the pensions, which in cash balance accounts can take the form of a lump sum rather than an annuity.

<sup>&</sup>lt;sup>8</sup> An example of how a Dutch collective pension fund would deal with a negative asset shock is provided by Bonenkamp, van de Ven and Westerhout (2007). For a general discussion of the costs and benefits of this type of pension schemes, see van der Lecq and Steenbeek (2007).

<sup>&</sup>lt;sup>9</sup> See Modigliani, Muralidhar and Ceprini (2000), Modigliani and Muralidhar (2004), and references therein.

<sup>&</sup>lt;sup>10</sup> Another advantage of the CFDB fund is that the pooled management of the contributions of all workers reduces the cost of managing the portfolio; also, workers with the same pension contributions and similar in all other respects are guaranteed similar rates of return.

### 3. Defined contribution pension schemes with forms of financial protection

DC pension funds may offer members either relative rate guarantees or fixed rate guarantees.<sup>11</sup> In the case of the former, the guaranteed return is tied to an index. A well-known example is Chile, where the guarantee is linked to the average return earned by all pension fund management companies. These forms of guarantee typically do not provide a hedge against generalized adverse conditions in financial markets. Fixed rate guarantees may take the form of a flat rate (such as 2.5 percent per year) or a minimum rate. In the case of the flat rate, the DC pension fund becomes a mixed DB pension scheme, like a cash balance plan. In what follows, therefore, we focus on minimum return guarantees.

Minimum return guarantees may take three basic forms: (1) they can be based on financial market instruments, as where portfolio choices are circumscribed to fixed-income securities (in particular those indexed to consumer prices) or derivatives options contracts are subscribed; (2) they can be provided by third parties, such as an insurance company or the pension fund sponsor; (3) they can be based on a reserve fund (at the level of the individual fund or groups of funds) or on other forms of mutual risk transfer.<sup>12</sup> The reserve funds, in particular, enable the financial results to be spread out over time: in years of high returns part of these are placed in the reserve fund, to be drawn on in years when the returns fall below a given threshold.

It must be stressed that these forms of protection always entail costs, unless the pension fund sponsor or another entity provides a subsidy. These costs may take the form of fees and premiums (for derivatives and insurance policies) or lower returns (when the accumulated amount is invested mainly in low-risk securities with intrinsically low returns or when the smoothing of returns over time penalizes some cohorts). If these lower net returns are not offset by higher contributions during the accumulation phase, they will result in smaller pensions. In other words, lower net returns always imply less consumption either during work life or in retirement or both.

Minimum return guarantees based on financial market instruments or on mutual assistance schemes have two major drawbacks. First, as mentioned, they may entail considerable costs. Second, and more important, they may not be able to insure against simultaneous, extreme, systemic shocks to a range of portfolio asset classes. In such adverse contingencies, it is very likely that the terms and conditions of the guarantees would be renegotiated.

These concerns are consistent with recent evidence on the functioning of minimum return guarantees. Turner and Rajnes (2009) analyze the cases of some pension funds in the United States<sup>13</sup> and in some other countries during the last two major stock market declines (in 2000-2003 and 2007-2009). They find that guarantee schemes in which the minimum return is set at a low rate, such as return of principal, seem to be able to withstand adverse financial market conditions. The results differ somewhat for minimum return guarantees above the risk-free rate, which tend to be expensive and to be maintained through subsidies by the provider. Moreover, over the

<sup>&</sup>lt;sup>11</sup> See Turner and Rajnes (2001, 2003, 2009) and Pennacchi (2002).

<sup>&</sup>lt;sup>12</sup> The latter case is considered by Boeri, Bovenberg, Coeuré and Roberts (2006).

<sup>&</sup>lt;sup>13</sup> Funds sponsored by state governments (for instance, Indiana, Louisiana, Montana, Ohio and Oregon) or non-profit organizations.

last decade expensive minimum return guarantees have often been changed as required, possibly by lowering the minimum return, denying the guarantee to new members, going over to a flat rate guarantee, or eliminating the guarantee outright.

### 4. A public minimum return guarantee for defined contribution schemes

How can the government insure participants in a defined contribution pension scheme against financial market tail risks, i.e. against very large declines in financial asset prices that are very rare but that can drastically reduce the value of a worker's accumulated pension fund assets? One way would be to provide a minimum return guarantee that protects future pensioners against any sharp curtailment of the amount accumulated. The minimum return should be set at a level that will (1) generate expenditure commitments for the government that can be financed from an ad hoc fund and that in any case are sustainable and (2) ensure each worker has an acceptable pension income in relation to the contributions paid in over his working life and his propensity to take investment risks.

One solution that appears reasonable is to make the guarantee compulsory and to offer DC plan members a set of options on the level of minimum return. Moreover, DC plan members should pay a risk-based premium for the guarantee, proportional to both the volatility of the portfolio and the level of minimum return.

Three reference choices for the minimum return are as follows: (*i*) a zero percent rate, which would ensure the return of contributions in nominal terms; (*ii*) the inflation rate, which would ensure the return of contributions in real terms; and (*iii*) the nominal growth rate of GDP. The latter choice resembles those adopted in the so-called notional pay-as-you-go public pension systems of some countries (for instance, Italy and Sweden).

In order to ensure the sustainability of the public insurance scheme, the guaranteed minimum return should not be greater than the nominal growth rate of GDP.<sup>14</sup> Minimum returns lower than the nominal GDP growth rate would be feasible, as plan members might want to exchange lower levels of coverage for lower premia. Therefore, provided that plan members would pay risk-based premia for the guarantee, any level of minimum return between nil and the nominal GDP growth rate could be appropriate, depending on DC plan members' willingness to bear investment risks. The rest of this section lays out how the government insurance programme would work.

The government would guarantee DC plan participants that upon retirement they would receive a final accumulated amount not lower than a "notional" amount given by the level that would have been reached if the contributions had been capitalized year by year at the guaranteed minimum return (e.g., the nominal growth rate of GDP):

$$W_T = \max(G_T, A_T) \tag{1}$$

<sup>&</sup>lt;sup>14</sup> In case of positive population growth, a minimum return equal to the growth rate of GDP per capita would likely ensure the stability of the long-run ratio between the total assets of pension schemes and GDP.

where  $W_T$  is the final amount to which the pension fund participant is entitled at the time of retirement and after T years of contribution,  $G_T$  is the final amount of the notional fund and  $A_T$  the final amount of the pension fund.

The final amount of the notional fund would be calculated as follows:

$$G_{T} = \sum_{e=e_{a}}^{e_{p}-1} c_{e} w_{e} \prod_{i=e}^{e_{p}-1} (1+g_{i})$$
(2)

where T is the number of years of contribution,  $e_p$  is the participant's age at retirement,  $e_a$  is his age at the time of enrolment in the pension fund, e is his current age,  $c_e$  is the percentage contribution rate,  $w_e$  is gross earnings and  $g_i$  is the guaranteed minimum return in the year in which the participant was *i* years old.<sup>15</sup>

The final amount of the pension fund can be expressed as:

$$A_{T} = \sum_{e=e_{a}}^{e_{p}-1} c_{e} w_{e} \prod_{i=e}^{e_{p}-1} (1+\rho_{i})$$
(3)

where  $\rho_i$  is the average rate of return earned by the DC plan on its portfolio of financial assets when the participant was *i* years old.

The guarantee, therefore, gives prospective retirees the right to enter into a swap of notional funds for pension funds with the government. If the guarantee is exercised, it implies a net transfer of resources from the government to the prospective retiree equal to the difference between the final amount of the notional fund and that of the pension fund:

$$\max(G_T - A_T, 0) = \max\left(\sum_{e=e_a}^{e_p-1} c_e w_e \left(\prod_{i=e}^{e_p-1} (1+g_i) - \prod_{i=e}^{e_p-1} (1+\rho_i)\right), 0\right).$$
(4)

In order to pay out this insurance cover, the Treasury could institute a special "guarantee fund", which would finance the operation by running down reserves or by issuing government-guaranteed securities. The profits would come from the premiums paid by DC plan participants to cover the costs of the guarantee and from income on financial operations.

The transfer of resources from the guarantee fund to the prospective retiree could take two different forms, depending on whether or not there is a transfer of assets from the pension fund to the guarantee fund:<sup>16</sup>

<sup>&</sup>lt;sup>15</sup> As argued above, three possible levels of minimum return are nil, the inflation rate and the nominal GDP growth rate. The guarantee might also be applied to only a part of the contributions (for example, those paid by the worker or, in the Italian case, the severance pay provisions allocated to the pension fund).

<sup>&</sup>lt;sup>16</sup> The two streams of payments may be either settled separately or netted. If they are netted, only one party (in this case, the guarantee fund) makes a payment, the net of the two streams, to the other. See Chance (2003).

Figure 1

### Stylized financial statements of the guarantee fund for defined contribution pension schemes (1)

BALANCE SHEET

Assets	Liabilities and net equity	Expenses	Revenues
financial assets	government- guaranteed	net outlays generated by swaps	premiums paid by DC plan members
(deficit)	retained profits	interest expense	interest, dividends and other proceeds on proprietary securities
(net loss for the year)	net profit for the year	valuation losses on proprietary securities	valuation gains on proprietary securities
		net profit for the year	(net loss for the year)

### INCOME STATEMENT

(1) The items "deficit" and "net loss for the year" appear in parentheses to indicate that they are alternative to the items "retained profits" and "net profit for the year", respectively.

- without transfer of assets, the guarantee fund only pays prospective retirees the difference between the final amounts of the notional funds and the pension funds (equation 4);
- with transfer of assets, the guarantee fund takes over the assets of prospective retirees (whose market value is given by equation 3) and pays them the full amount of their notional funds (equation 2); in this case the reserve fund has to finance much larger outlays and acquires the assets that the prospective retirees had accumulated with the pension funds.

In both cases, the stylized financial statements of the guarantee fund would be those presented in Figure 1. The income statement shows, under expenses, the net outlays generated by the swaps,<sup>17</sup> interest payments on government securities and valuation losses on proprietary securities. Under revenues, it would show premiums paid by DC plan participants, income on financial operations and valuation gains on the securities portfolio.

The balance sheet shows the sources of financing of the fund. Payments to prospective retirees would be financed by running down reserves or by proceeds from issues of government-guaranteed securities. Profits would be generated by the

<sup>&</sup>lt;sup>17</sup> Net outlays are defined as the difference between the notional funds and the market value of the assets accumulated into the pension funds, regardless of whether the swap entails a transfer of assets or not.

build-up of premiums paid by pension plan participants benefiting from the guarantee and by investment of those premiums in financial assets, as well as by income and capital gains on the financial assets.<sup>18</sup> An assessment of the level of the premiums that workers would have to pay is provided in Section 8.

### 5. Long-run growth in GDP and share prices: a comparison

For the sustainability of the guarantee fund, the most demanding option is a minimum return equal to the nominal growth rate of GDP. In that case, the public insurance scheme described above has two main factors of risk: the nominal growth rate of the economy and the returns of the financial markets. Before examining the advantages and costs of this form of guarantee for the workers, as well as the risks for the government, let us look at some empirical regularities in the relative performance of GDP and financial asset prices. Our analysis of portfolio returns focuses on equities, for which long time series are more available.

Some statistics on the long-run relation between GDP rates and equity returns are presented in Table 1. Panel (a) of the table considers data on the United States covering 80 years, from 1929 to 2008. For four long time horizons (10, 20, 30 and 40 years), the table shows the years in which the average growth rate of GDP was higher than the average rate of return on equities for at least one of the four horizons. The table highlights that, for long time horizons, the cases in which nominal GDP outpaces the stock index:

- are not very frequent. This happens in 20 of the 70 years for the 10-year horizon, 11 of 60 for the 20-year horizon and never for the 30- and 40-year horizons (for which 50 and 40 observations, respectively, are available);<sup>19</sup>
- are rarer, the longer the investment horizon;
- show some degree of persistence. In the 80 years examined, the cases in which the growth rate of GDP exceeds that of share prices are concentrated in two sub-periods: the years just after the Second World War and the 1970s-early 1980s.

Panel (*b*) of the table compares data for 10-year periods starting from 1970 for 15 countries. The main results are:

- the relative performance of GDP and equities shows marked similarity across countries. Almost everywhere the GDP growth rate exceeds that of equity prices in the 1970s, while the opposite occurs in the two subsequent decades;
- for the current decade, the data now available indicate that in all 15 countries the equity indices have gained less than GDP, owing to the collapse of the stock markets in 2008.

Broadly similar results (not reported) are obtained from a comparison of GDP growth with a portfolio consisting entirely of bonds and with a portfolio divided

<sup>&</sup>lt;sup>18</sup> Initially the fund's reserves will be very low, because workers will have been paying premiums only for a short period. In case the guarantee is triggered, the fund will run a deficit. With the passage of time, outpayments of notional funds will presumably cease, and, thanks also to income and valuation gains on the financial assets acquired with the premiums paid by plan members or with swaps (if there is a transfer of assets), the guarantee fund would begin to show a profit and make good the deficits (since at a certain point it would also be able to begin set aside reserves).

<sup>&</sup>lt;sup>19</sup> As the time series begins in 1929, the rates of return on periods of 10, 20, 30 and 40 years are available starting from 1939, 1949, 1959 and 1969, respectively.

### A comparison of long-run growth rates in nominal GDP and equity indexes (1)

(excess of the growth rate of GDP over that of share prices, in percentage points and on an annual basis)

Panel (a): Years in which the long-run annual growth rate in nominal GDP exceeded that of the stock price index in the US between 1929 and 2008 (2)

		Year →	42	43	44	45	46	47	48	49	71	73	74	75	76	77	78	79	80	81	82	83	84	85	08
	10-year gro	owth rates	1.62	6.80	3.38	3.50	6.32	2.06	4.37	2.15	0.17	2.87	6.92	4.77	3.13	6.18	6.04	3.84	2.47	4.97	3.40				6.80
	20-year gro	owth rates								0.55			0.01			0.16	1.73	1.45	0.81	2.56	0.87	1.07	1.66	0.95	
	30-year gro	owth rates																							1
	40-year gro	owth rates																							
								Pane	el (b)	: 15 adva	nced co	untri	es sin	ce 19	70 (3	)									
		Australia	Belg	gium	Canada	a De	enmark	Frar	nce	Germany	Ireland	1	Italy	Jap	an	, Nether	lands	Spa	in S	weden	Swit	tzerlan	d	UK	US
GDP [a	n																								
19	970-1979	13.0		10.4	12.	6	11.7		12.6	7.9	17.	6	17.0	1	12.3		10.7	1	7.9	11.0	1	6.	2	14.9	9.6
19	980-1989	11.0		6.4	8.	6	7.9		9.1	4.9	11.	3	13.6		6.1		4.2	1	2.3	9.9	1	5.	6	9.7	7.6
19	990-2000	5.0		4.2	4.	5	4.3		3.5	4.2	10.	2	5.7		2.0		5.4		7.2	4.9		2.	9	5.6	5.3
20	001-2007	7.1		4.1	5.	1	3.8		3.9	2.3	8.	5	3.7		0.4		4.4		7.3	4.4		3.	0	5.2	4.9
20	001-2008	7.3		3.9	5.	0	3.7		3.8	2.4	7.2	2	3.5		0.1		4.4		6.9	4.2		3.1		4.9	4.7
19	970-2000	9.6		6.9	8.	4	7.9		8.2	5.6	12.	9	11.8		6.5		6.7	1	2.2	8.4		4.	В	9.9	7.6
19	970-2008	9.1		6.3	7.	6	7.0		7.3	4.9	11.	7	10.0		5.2		6.2	1	1.1	7.5		4.	5	8.8	7.0
Equity [	[ <b>b</b> ]																								
19	970-1979	5.5		6.6	10.	0	9.6		8.3	2.6	13.	6	0.7	1	12.4		5.2	-	0.4	6.9	1	2.	5	11.7	6.7
19	980-1989	16.9		19.9	11.	7	23.6	2	22.3	16.8	22.	5	25.5	2	20.5		19.6	2	8.2	30.7		10.4	4	22.3	16.1
19	990-2000	10.7		10.2	10.	0	10.0		12.1	9.9	14.	2	10.3		-6.2		18.0	1	3.3	15.2		15.	6	12.7	14.4
20	001-2007	13.1		8.3	9.	0	9.6		4.4	4.7	5.	0	3.2		3.0		2.5	1	0.5	5.3		3.4	4	4.7	1.5
20	001-2008	5.4		-2.7	3.	0	0.2		-2.6	-2.1	-8.	5	-4.8		-3.8		-7.3		3.5	-1.6		-2.	1	-0.5	-4.7
19	970-2000	10.8		12.0	10.	5	14.0		14.0	9.5	16.	4	11.4		7.8		14.0	1	2.7	17.0	1	9.4	4	15.2	12.3
19	970-2008	9.7		8.8	8.	9	11.0		10.4	7.0	10.	8	7.9		5.3		9.3	1	0.7	12.9		6.	9	11.8	8.6
GDP vs	s. equity [a-b]																								
19	970-1979	7.5		3.8	2.	6	2.1		4.3	5.3	4.	0	16.3		-0.1		5.5	1	8.3	4.1		3.	7	3.2	2.9
19	980-1989	-5.9	-	13.5	-3.	1	-15.7	· _ ·	13.2	-11.9	-11.	2	-11.9	-1	14.4		-15.4	-1	5.9	-20.8		-4.	В	-12.6	-8.5
19	990-2000	-5.7		-6.0	-5.	5	-5.7		-8.6	-5.7	-4.	0	-4.6		8.2		-12.6	-	6.1	-10.3		-12.	7	-7.1	-9.1
20	001-2007	-6.0		-4.2	-3.	9	-5.8		-0.5	-2.4	3.	5	0.5		-2.6		1.9	-	3.2	-0.9	1	-0.4	4	0.4	3.4
20	001-2008	1.9		6.7	2.	0	3.5		6.4	4.5	15.	6	8.3		4.0		11.7		3.4	5.8		5.	2	5.3	9.4
19	970-2000	-1.2		-5.1	-2.	1	-6.1		-5.8	-3.9	-3.	5	0.4		-1.3		-7.3	-	0.5	-8.6		-4.	6	-5.3	-4.7
19	970-2008	-0.6		-2.5	-1.	3	-4.0		-3.1	-2.1	0.	9	2.1		-0.2		-3.1		0.3	-5.4		-2.	5	-3.0	-1.6

Sources: Based on Bureau of Economic Analysis, Bloomberg, OECD.Stat (http://stats.oecd.org/Index.aspx), Thomson Financial Datastream, Shiller (2005) and Dimson, Marsh and Staunton (2002).

(1) Equity indexes all incorporate reinvested dividends. – (2) Dashed vertical lines mark discontinuities in the sequence of years. – (3) Equity returns are only available for 10-year periods starting from 1970. For Belgium, Canada, Germany, Ireland, Italy, Japan, Spain, Switzerland and the UK, GDP growth rates are available since 1971.

equally between equities and bonds (the comparison is limited to the three sub-periods 1970-79, 1980-89 and 1990-2000).

Summing up, this section provides some evidence on the risk that over longterm horizons financial market returns turn out to be lower than the nominal growth rate of GDP. On the basis of past experience and available statistics, that risk has two main characteristics: (*i*) it is relatively low and tends to diminish as the investment horizon lengthens; (*ii*) it is, however, also poorly diversifiable, as it is highly concentrated in time and tends to affect many geographical areas simultaneously.

### 6. The advantages for the worker of a government-guaranteed minimum return

In the insurance scheme presented in Section 4, the guarantee would be compulsory, but plan members could opt among different levels of minimum return, ranging from zero to the nominal GDP growth rate. Such a guarantee scheme would offer several advantages to DC plan members.

In the first place, the contributions of each member during his working life would be capitalized at a rate that is at least equal to the level consistent with the maximum amount of risk that the worker is willing to bear (e.g., a rate that provides a final balance not lower than cumulative contributions adjusted for consumer price inflation). In this sense, a noteworthy case is a minimum return equal to nominal GDP growth, where for amounts saved under the DC plan during the accumulation period workers would not be exposed to inflation shocks or real shocks affecting single industries or groups of workers and would also be sheltered from the risk of becoming poorer in relation to the rest of the economy. A nominal GDP guarantee would also provide cover against protracted periods of unsatisfactory financial market returns (like the 1940s or 1970s), when the diversification of investments over financial asset classes and geographical areas does not significantly reduce risks (see Section 5).

Second, for workers, DC plans with a government-guaranteed minimum return would be better than either mixed DB plans or DC plans with financial protection (see Sections 2 and 3, respectively). Compared with mixed DB plans, DC schemes with a guaranteed minimum return give members the chance to benefit when the financial markets are performing well. If, at the end of the savings period, the return on the pension fund is greater than the accumulated growth of nominal GDP, the worker has the right to the sum accrued in the pension fund, whereas mixed DB plans have a predetermined capitalization rate for the contributions, and any surplus financial market returns go to the entities providing the guarantee.<sup>20</sup> Compared with DC plans with investment risk protection based on financial market instruments or on mutual assistance schemes, a publicly guaranteed minimum return may be less costly and more effective in addressing extreme financial risks (see also Section 9).

Lastly, the guarantee scheme presented in this paper would have the advantage of maintaining a high degree of flexibility in DC retirement schemes, above all in

<sup>&</sup>lt;sup>20</sup> The asymmetry of returns received by members of DC plans with a guaranteed minimum return (i.e. the fact that they benefit from positive trends in the financial markets without being exposed to the risk of falling asset prices) creates the risk of opportunistic behaviour by DC plan members. This is discussed in the next section.

terms of differentiating accumulation plans according to members' risk preferences and to different options for disbursing pension benefits.<sup>21</sup>

It must be made very clear that all these advantages would not be free. As we mentioned in Section 4, workers should pay a fee that takes account both of the riskiness of the portfolio and of the level of the minimum return (for an estimate of the market value of such a fee, see Section 8). This implies that the advantage of a public minimum return guarantee for defined contribution pension scheme members depends critically on the broader retirement system, and specifically on the relative size of defined contribution schemes within that system, the riskiness of the other components, whether enrollment in private DC schemes is mandatory or voluntary, and the standards of pension fund investment regulation. Therefore, possible applications of the guarantee scheme set out in this paper would have to be adapted to the specific characteristics of the arrangements applying in any one jurisdiction. It should also be pointed out that the variability of the cost of the guarantee implies that the guarantees that are more valuable are also those less affordable for low-income workers.

It is also worth noting that the guarantee would only apply to the accumulation phase, leaving open the question of how to deal with longevity and other risks and ensure that retirement wealth is properly decumulated.<sup>22</sup> In order to shield workers from investment and longevity risks during the payout phase, the minimum return guarantee should be combined with other options, such as postponing retirement or purchasing deferred annuities.

The guarantee might apply to both mandatory and voluntary DC pension schemes. Workers who voluntarily join a DC pension scheme and have a high propensity to take financial risks might always choose the 0 percent minimum return guarantee, in order to keep insurance premia as low as possible.

Another issue relates to the length of the contribution period. Since the latter is not known in advance, at least in voluntary systems, it could be possible to introduce a default length of the guarantee depending on the age of the worker at enrolment. That should be coupled with an additional charge if the worker exits earlier or a rebate if he remains in the fund longer.

Finally, the introduction of a compulsory choice among different levels of guaranteed minimum return might add a further layer of complexity to decisions about retirement, which typically turn out to be very difficult for workers. This fact might call for the introduction of a default option in the guarantee scheme. Any default option for the guarantee should be coordinated with possible default options for the portfolio. If there is no default option for the portfolio, the default guarantee could be a guarantee of repayment of contributions paid in in real terms (e.g., a 2.5 per cent guarantee). If there is a default option for the investment line, the default guarantee would depend on the latter. If the default portfolio were very low-risk, then

<sup>&</sup>lt;sup>21</sup> This insurance scheme may be compared to the innovative retirement system proposed by Kotlikoff (2008). Kotlikoff advocates replacing mixed retirement systems (including public and private pensions) with a personal account system characterized by: (1) compulsory contributions (with government matching contributions for the poor); (2) compulsory annuitization in real terms at retirement; (3) collective investment in a global, market-weighted index fund of stocks and bonds, subject to a guarantee of at least a zero real return on one's cumulative contributions.

<sup>&</sup>lt;sup>22</sup> On these issues see, e.g., Fornero and Luciano (2004), Visco (2007, 2009) and Guazzarotti and Tommasino (2008).

the default guarantee should be the 0 percent rate. If the default portfolio were a balanced portfolio, the default level of the guaranteed minimum return could be higher.

### 7. The risks of a guarantee fund

A guarantee fund exposes the government to two risks in particular: the uncertain sustainability of the insurance scheme and the possibility of opportunistic behaviour by the insured. Specific problems also arise in relation to the transition period and the governance of the guarantee fund. The discussion that follows focuses on a minimum return equal to the nominal growth rate of GDP, which implies the largest expected outlays for the guarantee fund.

*SUSTAINABILITY OF THE GOVERNMENT GUARANTEE SCHEME.* As regards the sustainability of the insurance scheme,<sup>23</sup> the first risk factor is financial market trends: the greater the volatility of the return on the pension fund portfolio, the greater the likelihood that the guarantee will be triggered and the greater the government's potential outlay. The second risk factor is the rate of GDP growth, which has direct and indirect effects working through the growth in gross earnings. The higher the rate of GDP growth, the higher the probability that the guarantee will be triggered and, in that case, the larger the net outlays of the guarantee fund.<sup>24</sup>

In order to assess the sustainability of the government guarantee scheme, it is worth observing that, in the long term, the returns on financial assets tend to be at least equal to the nominal rate of growth of the economy (see the evidence in Section 5). Moreover, as a government-backed entity, the guarantee fund would be able to finance the swaps with prospective retirees by issuing very long-term bonds at low interest rates.

A broad indication of how long it takes for the guarantee fund to recover the net outlays originated by the swaps can be gained from long time series on the United States. As we saw in Panel (*a*) of Table 1 in Section 5, over the period 1929-2008 there were some years in which, for accumulation periods of 10 and 20 years, nominal GDP outpaced the stock market. Those years are considered in Figure 2, which shows, for each of the two investment horizons, the distribution of the number of years it took for the share price index to regain the level of nominal GDP after the swap was executed.<sup>25</sup> The estimated recovery periods do not exceed 6 and 3 years for the 10- and 20-year horizons, respectively. If one also takes into account possible funding costs incurred by the guarantee fund (proxied by the annual nominal growth

<sup>&</sup>lt;sup>23</sup> The guarantee fund's outlay produced by the swap with pension fund members would be positively correlated with three variables (see equation 4 above): (*i*) the rate of GDP growth; (*ii*) gross earnings; and (*iii*) the percentage contribution rate. It would be negatively correlated with: (*iv*) the return on the DC plan. In the event the swaps were to be funded by issuing state-backed securities, the overall cost of the guarantee would also be affected by the rate of interest on these securities.

<sup>&</sup>lt;sup>24</sup> The size of the outlays is also affected by the contribution rates: even a variation of just a few tenths of a percentage point may have a significant impact, if applied to the entire amount of annual earnings.

<sup>&</sup>lt;sup>25</sup> The indicator shown in Figure 2 does not take into account contributions, which have an impact on the time to recovery, while it does take into account very few cases of a second drop of the share price index below the value of the guarantee. With respect to the cases shown in Panel (a) of Table 1, the left-hand panel of Figure 2 does not include 2008, for which post-shock returns are not yet available.

### Figure 2

### Indicator of the time to recover net outlays of swaps (1)

Swaps at the end of a 10-year accumulation period

Swaps at the end of a 20-year accumulation period



I recovery of the net outlay that had to be funded when the swap was executed recovery of the net outlay, if the latter is capitalized year by year at the nominal growth rate in GDP

### Sources: Based on Bureau of Economic Analysis and Bloomberg data.

(1) Data refer to the United States and cover 80 years, from 1929 to 2008. The two panels relate to 10-year or 20-year accumulation periods and show the distribution of the number of years after the swap in which the share price index was below the value it should have had in order for the guarantee not to be exercised. Two different hypotheses are made on the amount that has to be recovered: only the net outlay at the time of the swap and the net outlay capitalized year by year at the nominal growth rate of GDP.

rate of GDP), the recovery period naturally lengthens, reaching a maximum of 14 and 7 years for the 10- and 20-year horizons, respectively.<sup>26</sup>

The indicator reported in Figure 2 suggests that the time that is necessary to recover the net outlays of the swap is quite variable: it may be as short as one year, but it may also be as long as three years or more. This variability is a very serious risk for workers nearing retirement age, who may not be able or willing to postpone retirement for an indefinite number of years. But that uncertainty could be quite easily managed by the public guarantee fund, which would be a government-backed intermediary with a long-term investment horizon and able to issue very long-term bonds economically.

It will be helpful to estimate the outlays that the guarantee fund would have sustained over the last two decades if an insurance scheme of the kind proposed in this paper had been in place. The last two decades represent a tough test for such a fund. As shown in Section 5, in 2008 the financial crisis made nominal GDP outpace equity indexes over long-term investment horizons for the first time since the early eighties. Another period of dismal performance of stock markets in all advanced economies was right at the turn of the century, due to the bursting of the high-tech bubble. The last two decades also cover the most challenging test one can think of, that of the Japanese stock market. Since the major crash that occurred in 1990, share price indices in Japan have not yet fully recovered their end-1989 level and even in the middle of '00s (before the last crisis) they were still considerably lower than that level.

<sup>&</sup>lt;sup>26</sup> If funding costs are proxied by the interest rate on long-term US government bonds (data drawn from Shiller, 2005), the maximum length of the recovery period is equal to 9 and 8 years for the 10- and 20-year horizons, respectively.

Figure 3 shows some estimates of the maximum outlays of the guarantee fund for the United States and Japan (Appendix A details the assumptions underlying the calculations), positing that the guarantee fund settles the net amount due in cash (equation 4), with no transfer of assets. The calculations provide an upper bound, as they are based on an extremely unfavourable hypothesis: all workers who retired in a given year had enrolled into a DC pension plan fully invested in equities and had made the worst choice as to the year of enrolment. For instance, with the benefit of hindsight, for a US worker who retired in 2008 the worst choice would have been to join a DC plan in 1996 and keep the money invested in an equity portfolio throughout the 13-year accumulation period. Figure 3 shows that if all US workers retiring in 2008 had made that choice, the estimated outlay for the guarantee fund would have amounted to 0.4-0.5 percent of 2009 GDP. For Japanese workers retiring in 2008 the worst choice would have been to join a DC plan in 1983 and hold a domestic equity portfolio for 26 years. If all Japanese workers retiring in 2008 had made that choice, in that year the guarantee fund would have transferred to prospective retirees a net amount of resources equal to about 0.7 percent of 2009 GDP. Note that these estimates assume that each year all prospective retirees were enrolled in a DC pension plan. This is an extreme assumption. In the United States, for instance, in 2008 only one senior worker out of two was enrolled in a DC plan (either occupational or individual; see Copeland, 2008).

Overall, the estimates reported in Figure 3 indicate that in worst-case scenarios the annual swap outlays of the guarantee fund might be as high as 0.7 percent of GDP. Figure 3 also shows an estimate of the cumulative annual interest expenses of the guarantee fund had it funded the maximum swap outlays by issuing 10-year government-backed bonds. Over the last two decades, the overall interest expenses would have amounted to 0.05 percent of the 2009 GDP in Japan and 0.03 percent in the United States.

The calculations shown in Figure 3 indicate that the swap outlays of the guarantee fund would likely be concentrated during periods of cyclical downturns and prolonged decline in stock market indices. This highlights the importance of prefunding the swaps by accumulating reserves. Prefunding would also shield the insurance scheme from political pressures.

If there were also a transfer of assets from the pension schemes to the guarantee fund, the swap outlays would be much larger and the guarantee fund would likely have higher leverage, i.e. a higher level of debt for given reserves. Higher leverage might allow the guarantee fund to more quickly recoup swap outlays. On the other hand, the transfer of assets would make the management of the swaps more complex and would increase both the solvency risk of the guarantee fund and other possible disadvantages entailed by size (e.g., the risk of suggestions to use reserves for purposes different from that of providing the guarantee).

In any case, the guarantee fund's size compared with the size of the main investment markets should be monitored to make sure that financial asset prices are not excessively influenced by the fund's portfolio choices. Figure 4 compares the total assets of pension funds with the size of the domestic bond and equity markets in three countries where retirement saving schemes have long been developed. While it would be necessary to take into account the geographical diversification of pension fund holdings, this simple comparison suggests that the guarantee fund may get large enough to have a substantial price impact on some financial assets.

### Figure 3

## Estimate of the maximum outlays of the guarantee fund for the United States and Japan if the minimum return guarantee scheme had been in place (1)

(as a percent of the GDP of the following year)



Sources: Based on national data and Bloomberg, OECD, Shiller (2005) and Thomson Datastream data. (1) For each country, estimate of the maximum annual outlays (and cumulated interest expenses) of the guarantee fund had the minimum return guarantee scheme been in place (as a percent of the GDP of the year following the one shown on the horizontal axis). It is assumed that the guarantee fund settles the net amount due in cash with no transfer of assets. The maximum outlays are estimated under the extremely unfavourable assumption that, in each year, all workers who retired in that year: (*i*) had previously joined a DC pension plan, kept their balances invested in 100 percent domestic equity portfolios and opted for a nominal GDP guarantee; (*ii*) had all joined the plans in the worst year in terms of the return of the stock market index relative to nominal GDP over the accumulation period. For each year in which there are outlays of the guarantee fund, the graph also reports the hypothetical year in which that cohort of new retirees joined the plans and the length of the accumulation period in years. The annual contribution rate is set equal to 12 percent and the annual number of new retirees is estimated on the basis of the number of workers who were 60-64 years old. Due to data availability, the counterfactual analysis is carried out for any possible accumulation period starting from the early nineties for the United States and the early eighties for Japan. For a description of the data, see Appendix A.

THE TRANSITION PERIOD. The countercyclicality of swap net outlays calls for special attention during the transition period. It would be better to limit the size of interventions until the guarantee fund had built up a certain level of reserves. That could be done by establishing that the guarantee only applies to accumulation periods longer than a certain number of years (say, fifteen), to be shortened gradually as premia accrue. The time of activation of the guarantee might also be varied depending on the riskiness of the portfolio and the level of the guarantee.

RISKS OF OPPORTUNISTIC BEHAVIOUR BY THE INSURED OR THE ASSET MANAGER. The existence of a government-guaranteed minimum return could encourage DC plan members to over-expose themselves to financial risks. This risk could be a significant one especially for workers who, as they near retirement, are dissatisfied with the results achieved by their pension fund. A return guarantee might also affect the incentives of portfolio managers. Opportunistic portfolio choices could be discouraged in several ways. First, the guarantee scheme should be made compulsory, especially if members have to pay regular premiums. Second, the guarantee might be applied only if the period of participation in the DC scheme is longer than some threshold duration (say, ten years). These two conditions would reduce adverse selection effects. Third, a ceiling on the percentage of risky securities in a pension fund's portfolio could be set. Fourth, the allocation of securities within a portfolio

### Pension fund holdings compared with the size of domestic market, 2005



(pension fund total assets/market capitalization)

could be required to have an individual "life cycle"; that is, as retirement approaches, the portfolio would be redistributed away from risky securities to safer assets (e.g. from shares to government securities). Fifth, other prudential criteria could be tightened, such as those meant to ensure the diversification of risk within the portfolio and the quality of eligible portfolio assets. Finally, following Lachance and Mitchell (2003a, 2003b), the minimum return guarantee could be applied to a predefined standard portfolio (e.g. monetary, balanced, growth), rather than to the portfolio actually held by the prospective retiree. In brief, risks of opportunistic behaviour by DC plan members or portfolio managers could be mitigated by sound regulation of workers' options and pension fund investments.

*MANAGEMENT OF THE GUARANTEE FUND.* Specific problems concerning the government guarantee fund could arise from the management of the financial assets portfolio. A solution could be to entrust the management to specialized intermediaries, appointed under transparent and competitive procedures. The investment objective could be a long-term return at least equal to that on government securities (or the rate guaranteed with the swap), plus a spread of one or two percentage points in order to recoup the net outlays entailed in the swap. Voting rights at the meetings of shareholders of listed companies should be adequately regulated. In brief, the guarantee fund should put in place governance mechanisms and investment controls to ensure sound management and better insulate both governance and asset management from undue political influence. A way to achieve this would be to follow the international best practices that have been identified by the OECD and other bodies for public pension reserve funds (Yermo, 2008).

### 8. An assessment of the premium payable by the pension scheme members

The final amount to which the member is entitled at the time of retirement (equation 1) can be rewritten as:

$$W_T = A_T + \max(G_T - A_T, 0).$$
 (5)

The second term on the right-hand side of equation (5) is the payment generated by a put option whose strike price is  $G_T$ . Consequently, at retirement the worker is entitled to the amount accumulated with the pension fund  $(A_T)$ , supplemented by the payment generated by the put option. In each year *t* the worker holds the portfolio of financial assets accumulated with the pension fund up to that time  $(A_t)$  and the European put option offered to him by the government. Denoting the value of the option at time *t* by  $f_t$ , the total value of the position accrued by the member at that date is:

$$W_t = A_t + f_t \,. \tag{6}$$

Option value theory makes it possible to determine the price  $f_t$  of this option. Table 2 shows the results obtained with this purely financial approach.<sup>27</sup> To take proper account of the possibility of large drops in share prices, these are assumed to follow a jump-diffusion process. Moreover, the parameters of the process, which we take from Di Cesare (2005), are inferred from option prices over a period of high stock market volatility.<sup>28</sup> In this way we guard against the risk of underestimating the costs of the nominal GDP guarantee. The cost estimates shown in Table 2 should provide an upper bound on the market value of the guarantee, based on historical data on option prices.

Three different types of guarantee are considered in the simulations: two of them guarantee the repayment of the contributions paid in, in nominal and real terms (minimum rates of return of 0 and 2.5 percent respectively).<sup>29</sup> The third type of guarantee is that discussed in Section 4, in which the minimum rate of return of the pension fund is not predetermined but put equal to the nominal GDP growth rate in each year.<sup>30</sup> As regards the riskiness of the portfolio, four investment strategies are considered: (*i*) 100% equities; (*ii*) 50% equities and 50% government securities; (*iii*) 100% government securities; and (*iv*) a simple life-cycle strategy, under which the portfolio is initially invested 100% in equities and then, in the last ten years of the accumulation period, gradually shifts, on a linear basis, into government securities

<sup>&</sup>lt;sup>27</sup> The fair value of the option is based on risk-neutral valuation. The methodology, and the underlying hypotheses, are presented in Appendix B. The solution can be obtained numerically, as the use of closed formulas  $\dot{a}$  la Black and Scholes is hindered by the presence of periodic investments (see, for example, Zurita, 1994, and Pennacchi, 1999). For a thorough analysis of the main determinants of the cost of guaranteeing a minimum rate of return for a pension fund, see Lachance and Mitchell (2003a, 2003b).

<sup>&</sup>lt;sup>28</sup> The jump-diffusion process is driven by three parameters: continuous volatility, the expected number of jumps per year, and the average jump size (see Appendix B). They are based on the prices of the options on S&P 500 futures contracts listed on the Chicago Mercantile Exchange (Di Cesare, 2005). The estimation period runs from August 2000 to July 2002 and is characterized by quite high share price volatility, associated with the collapse of the high-tech bubble, a recession, 9/11, and accounting frauds.

<sup>&</sup>lt;sup>29</sup> In regimes of low inflation a rate of return of 2.5 percent is sufficient to ensure repayment of the principal in real terms.

<sup>&</sup>lt;sup>30</sup> It is important to note that, owing to the assumption of risk neutral valuation required to determine the price of the option, for all the risky assets, including nominal GDP, the rate of return expected by investors coincides with the risk-free interest rate. This means that no assumption is ever made in the simulations regarding the equity premium, the term premium or expected nominal GDP growth. On the other hand, the assumptions regarding the volatilities of the various stochastic processes considered and their correlations are of crucial importance.

#### Years with Individual Account Minimum Rate of Return 0 percent 2.5 percent Yearly nominal GDP growth rate (2) (i.e. return of the principal) (i.e. about long-run inflation) I. Portfolio invested 100% in equities with jumps (3) without jumps (3) 10 1.47 2.39 3.04 2.57 20 0.68 1.46 2.14 1.82 30 0.41 1.07 1.77 1.48 40 0.26 0.83 1.54 1.27 II. Portfolio invested 50% in equities and 50% in 10-year Treasury bonds 10 0.35 0.97 1.54 20 0.11 0.54 1.13 30 0.05 0.37 0.95 40 0.02 0.26 0.81 III. Portfolio invested 100% in 10-year Treasury bonds 10 0.00 0.00 0.44 20 0.00 0.00 0.32 30 0.00 0.00 0.27 40 0.00 0.00 0.23 IV. Life-cycle investing (4) 10 0.08 0.52 0.93 20 0.09 1.24 0.56 30 0.10 0.59 1.22 40 0.03 0.34 1.20

### Cost Estimates of Minimum Return Guarantees for Defined Contribution Pension Schemes (1) Panel (*a*): annual charges as a percentage of total assets

Source: Authors' calculations (see Appendix B).

(1) Estimate of the cost of a European put option with the maturity shown in the first column. Estimates are based on Monte Carlo simulations (10,000 trials) and are carried out for four different portfolios. As for the definition of minimum return, three different formulas are considered: two fixed-rate guarantees (0 and 2.5 percent per annum) and a guarantee yielding the nominal GDP growth rate each year. The methodology and underlying hypotheses are presented in Appendix B. - (2) The volatility of the nominal GDP growth rate is set equal to 2 percent per annum; the correlation between the GDP growth rate and stock returns is set equal to 0.4. - (3) For the case of a nominal GDP guarantee on a portfolio invested all in equity, the table also shows the cost of the guarantee if share prices do not follow a jump process. - (4) The portfolio is initially invested in equities and then linearly switches into bonds in the 10 years prior to retirement.

Years with Individual Account	Minimum Rate of Return							
	0 percent (i.e. return of the principal)	2.5 percent (i.e. about long-run inflation)	Yearly nominal GDP growth rate (2)					
	I. Portfe	plio invested 100% in equities						
10	11.40	18.48	23.49					
20	9.04	19.25	28.30					
30	7.69	19.98	33.12					
40	6.39	20.05	37.17					
	II. Portfolio invested 50%	in equities and 50% in 10-year Tre	easury bonds					
10	2.68	7.42	11.81					
20	1.45	7.12	14.74					
30	0.84	6.83	17.45					
40	0.48	6.22	19.27					
	III. Portfolio inv	ested 100% in 10-year Treasury bor	nds					
10	0.00	0.00	3.33					
20	0.00	0.00	4.12					
30	0.00	0.00	4.89					
40	0.00	0.00	5.53					
	IV	. Life-cycle investing (3)						
10	1.89	12.37	28.71					
20	1.77	10.40	22.85					
30	1.34	7.76	16.26					
40	0.22	2.65	7.16					

### Cost Estimates of Minimum Return Guarantees for Defined Contribution Pension Schemes (1)

Panel (b): annual charges as a percentage of lifetime contributions

Source: Authors' calculations (see Appendix B).

(1) Estimate of the cost of a European put option with the maturity shown in the first column. Estimates are based on Monte Carlo simulations (10,000 trials) and are carried out for four different portfolios. As for the definition of minimum return, three different formulas are considered: two fixed-rate guarantees (0 and 2.5 percent per annum) and a guarantee yielding the nominal GDP growth rate each year. The methodology and underlying hypotheses are presented in Appendix B. - (2) The volatility of the nominal GDP growth rate is set equal to 2 percent per annum; the correlation between the GDP growth rate and stock returns is set equal to 0.4. - (3) The portfolio is initially invested in equities and then linearly switches into bonds in the 10 years prior to retirement.

until, at the time of retirement, the equity share is nil.<sup>31</sup> Lastly, four different investment durations are considered (10, 20, 30 and 40 years). GDP volatility is taken to be 2 percent, while the correlation between GDP and the pension fund portfolio is assumed to be 0.4.<sup>32</sup> The cost of the option is calculated on an annual basis and both as a percentage of the assets under management (Panel *a* of Table 2) and as a percentage of lifetime contributions (Panel *b*). The second measure provides an indication of the affordability of the guarantee, in terms of its incidence on DC plan members' disposable income.

The estimates show that the type of guarantee provided has a substantial effect on the cost of the option. If we consider a 40-year accumulation period and a balanced portfolio divided equally between equities and 10-year government bonds, the cost of the option is very low if the option guarantees only the repayment of the contributions paid in; it rises to 0.26 percent of assets under management (6.22 percent of contributions) when the option guarantees an annual nominal rate of return of 2.5 percent and 0.81 percent (19.27 percent) when it guarantees the nominal GDP growth rate.

Another factor that has a considerable influence on the cost of the option (measured in terms of assets or contributions) is the volatility of the portfolio. If the portfolio is fully invested in equities, even just guaranteeing repayment of the contributions paid in requires, over a 40-year time horizon, an annual premium equal to 0.26 percent of the assets under management, which rises to 1.54 percent if the strike price of the option is linked to the GDP growth rate (6.39 and 37.17 percent, respectively, in terms of contributions). Conversely, the cost of the option falls drastically if the percentage of equities in the portfolio is reduced to zero; it is nil in the case of repayment of the capital paid in (in nominal and real terms), and even where the amount guaranteed is linked to the nominal GDP growth rate it is still barely above 0.20 percent of total assets (5.53 percent of contributions). A life-cycle strategy tends to be less expensive (in terms of total assets) than the balanced strategy (50% equities and 50% securities) for the 10-year maturity and more expensive for the longer durations (because of the larger weight of the period in which the portfolio is fully invested in equities).

The presence of jumps in share prices implies quite a sizable increase in the cost of the guarantee, although its incidence tends to diminish as the accumulation period lengthens. This is shown in the last column of Table 2.*a*. For a portfolio fully invested in equities, the annual surcharge due to expected jumps in share prices is about 0.4 percent of assets under management for a 10-year time horizon and 0.2 percent for longer horizons.

Another variable that has a significant influence on the value of the guarantee is the duration of the accumulation period. If costs are measured as a percentage of assets under management, the option becomes less expensive as the investment horizon lengthens; the relation is markedly non-linear for relatively short retirement

<sup>&</sup>lt;sup>31</sup> The example of life-cycle investing is taken from Blake, Cairns and Dowd (2008).

<sup>&</sup>lt;sup>32</sup> The historical data on nominal GDP provide estimates of its volatility that are very low. On the basis of quarterly data for the United States covering the period from 1988:Q1 to 2008:Q4, it can be estimated that the volatility of nominal GDP is of the order of 1 percent on an annual basis. The correlation between nominal GDP and the S&P 500 index (with dividends reinvested) would not exceed 0.4.

investment horizons.<sup>33</sup> This negative correlation with the investment horizon does not generally hold if costs are expressed as a share of contributions, because this cost measure does not take into account the growth of assets due to the proceeds of the investment portfolio. However, if the guaranteed minimum return is considerably lower than the rate of growth of earnings, the fact that the guarantee becomes less generous over time may dominate the volatility effect and the guarantee cost may fall over time.<sup>34</sup> In Table 2.*b*, this holds for the guarantee of repayment of the contributions paid in in nominal terms and for the 2.5 percent flat rate guarantee on the balanced portfolio.

As for the guarantee linked to the nominal GDP growth rate, an important property that needs to be highlighted is that the amount insured (i.e.,  $G_T$  the strike price of the put option in equation 5) is itself a random variable, because it depends on the performance of GDP. This risk factor is an additional source of volatility of the derivative contract and as such should increase the value of the option. However, it can also reduce its value if the risk factor shows a significant positive correlation with the portfolio. It is therefore important to assess the sensitivity of the price of the option both to the volatility of GDP and to the latter's correlation with the portfolio. The estimates are shown in Table 3.<sup>35</sup> The simulations assume a balanced portfolio divided equally between equities and government securities, with a volatility on the order of 10.5 percent.<sup>36</sup> For the volatility of nominal GDP the values considered are between 1 and 10 percent, while for the correlation between nominal GDP and the portfolio, it is assumed to be positive and capable of taking on values ranging from 0.2 to 0.99 (almost perfect correlation).

The simulations show that a higher correlation between the value of the pension fund and that of GDP reduces the value of the option, both because it decreases the portion of cases in which the guarantee is triggered (i.e. in which the option gives rise to positive payoffs) and because it diminishes the size of the payoffs. The effect of the correlation depends, however, on the level of volatility of the two stochastic processes and, under standard assumptions regarding the volatility of GDP, it is relatively small. For example, an increase in the correlation from 0.4 to 0.6 reduces the value of the option by between 0.05 and 0.12 percent if the volatility of GDP is 4 percent and by between 0.10 and 0.23 percent if it is 6 percent.

As for the effect of an increase in the volatility of GDP on the value of the option, it is not unambiguous because it depends on the degree of correlation between GDP and the portfolio of the pension fund. An increase in the volatility of GDP has two effects of opposite sign: it amplifies the correlation effect referred to above, which lowers the value of the option, and has a direct effect on the overall volatility of the option, which increases the value of the derivative contract. As long as the volatility of GDP is relatively low, an increase in it will have little impact on the overall volatility of the option, while it strengthens the effect, of the opposite sign, of

<sup>&</sup>lt;sup>33</sup> See also Figure 4 in Cesari, Grande and Panetta (2008), in relation to a guarantee of repayment of the contributions paid in.

<sup>&</sup>lt;sup>34</sup> See a related discussion in Lachance and Mitchell (2003a). See also Bodie (1995, 2008).

<sup>&</sup>lt;sup>35</sup> For the derivation of these effects in closed form, under the simplifying hypotheses that there is a single contribution to the pension fund at the beginning of the accumulation period and that the share prices follow a Brownian geometric motion without jumps, see Appendix B.

<sup>&</sup>lt;sup>36</sup> As a result of standard assumptions on the volatilities of stock returns and short-term interest rates. See Appendix B.

Volatility of	Correlat	tion between no	ominal GDP an	d pension fund a	issets
nominal GDP (%)	0.2	0.4	0.6	0.8	0.99
			10 years		
1	1.38	1.32	1.25	1.24	1.21
2	1.36	1.29	1.24	1.15	1.11
4	1.39	1.26	1.14	0.99	0.84
6	1.43	1.31	1.08	0.87	0.59
8	1.58	1.36	1.14	0.85	0.38
10	1.76	1.52	1.26	0.89	0.28
			20 years		
1	1.00	0.94	0.91	0.91	0.90
2	0.95	0.94	0.90	0.85	0.79
4	0.99	0.89	0.82	0.73	0.61
6	1.05	0.95	0.80	0.64	0.42
8	1.13	0.99	0.81	0.61	0.27
10	1.27	1.06	0.91	0.65	0.21
			30 years		
1	0.81	0.78	0.77	0.74	0.74
2	0.78	0.76	0.74	0.70	0.65
4	0.81	0.75	0.66	0.60	0.49
6	0.86	0.77	0.65	0.52	0.36
8	0.90	0.82	0.68	0.50	0.22
10	1.04	0.89	0.74	0.52	0.17
			40 years		
1	0.71	0.69	0.68	0.64	0.64
2	0.69	0.66	0.64	0.61	0.57
4	0.69	0.65	0.60	0.51	0.43
6	0.76	0.67	0.57	0.46	0.31
8	0.80	0.71	0.59	0.44	0.20
10	0.91	0.78	0.64	0.45	0.15

### **Cost estimates of minimum return guarantees: sensitivity to different assumptions on GDP volatility** (1) (annual charges as a percentage of total assets)

Source: Authors' calculations (see Appendix B).

(1) Estimate of the cost of a European put option with a strike price linked to the yearly nominal GDP growth rate. The estimates are carried out for a number of combinations of GDP volatility and GDP-pension fund asset correlations, as well as for four different investment horizons (10, 20, 30 and 40 years). Risk-neutral valuation is assumed. The pension fund portfolio is evenly split between stocks and 10-year Treasury bonds and its volatility is equal to about 10.5 percent per annum. The estimates are based on Monte Carlo simulations (10,000 trials). For simplicity, share prices are assumed to follow a Brownian geometric motion without jumps. The methodology and the other underlying hypotheses are presented in Appendix B. The shaded areas show the combinations of GDP volatility/correlation for which, according to the pricing model in Margrabe (1978), in the case of a single contribution paid into the pension fund an increase in GDP volatility would lead to a rise in the value of the option (see Appendix C).

the correlation. Beyond a certain threshold, the direct effect exceeds the correlation effect and a further increase in the volatility of GDP produces an increase in the value of the option.

It is important to also highlight that the cost estimates reported in Table 2 assume that the guarantee's cash flows can be exactly replicated by trading financial assets. This is a restrictive assumption, as GDP-linked securities usually are not available and financial markets are likely to provide only an imperfect hedge against fluctuations of aggregate economic activity.<sup>37</sup> An assessment of how much the fair value of the option is affected by the lack of GDP-linked securities can be carried out under the hypothesis that the residual GDP risk (i.e. the risk that cannot be hedged by traded assets) is not priced by financial markets. Some estimates indicate that the costs of the nominal GDP guarantee might be slightly above the level shown in the last column of Table 2 (of about 0.1 percentage points).<sup>38</sup>

Overall, these simulations indicate that, if a purely financial approach is adopted (assuming risk-neutral valuation), then the cost of guaranteeing a minimum rate of return equal to the nominal GDP growth rate may be significant if the portfolio mostly include equities, even in the case of investment horizons stretching over decades.

However, for any given type of guarantee, the cost of the guarantee may be limited by introducing restrictions on portfolio risk, in the form, say, of limits on the amount invested in equities held (possibly applied in particular to the years preceding retirement). Alternatively, for any given level of portfolio risk, the cost of the guarantee may be lowered by reducing the guaranteed amount, by, for instance, ensuring the return of the principal only (in nominal or real terms).

### 9. Why should the guarantee be provided by the government?

The option pricing approach used in Section 8 gives an estimate of the market value of the guarantee, i.e. of the marginal cost that investors would charge the government to take over its commitments (see Pennacchi, 2002). This price can be determined under the hypotheses of complete markets (i.e. the assets traded can replicate the payoffs of the option) and no arbitrage opportunities (i.e., assets are traded continuously and there is no "free lunch"). In this framework, considering the nominal GDP guarantee, the guarantor (public or private) would always be able to hedge its exposure to it perfectly by trading the underlying securities. For example, if the insured portfolio were fully invested in a share price index, the guarantor could hedge by selling the index short and buying the nominal GDP index, so that when the

<sup>&</sup>lt;sup>37</sup> While the macro markets posited by Shiller (1995) are meant to overcome the difficulties faced by capital markets in hedging aggregate income risks, Kamstra and Shiller (2009) make the case for government issues of securities with coupons tied to nominal GDP. See the discussion in the next section. For estimates of bounds on prices of non-traded assets, see Cochrane and Saá-Requejo (2000) and Kaido and White (2009).

<sup>&</sup>lt;sup>38</sup> The methodology has two stages (see Pennacchi 1999, 2002): (*i*) first, estimate the difference between the expected growth rate of GDP and the equilibrium expected return of a hypothetical security that has the same volatility of GDP; (*ii*) when running Monte Carlo simulations, in the drift of the GDP process add that difference to the risk-free rate. In the estimates, the expected GDP growth rate is set equal to its annual average. The expected return on a hypothetical GDP-linked security is estimated through an asset pricing model including three factors (stock market returns, 10-year Treasury bond yields and consumer price inflation). Estimates are carried out recursively on annual data from 1980-2006 to 1991-2006 and, in each iteration, the difference between the sample average of the GDP growth rate and the expected return is computed. The average value of this difference is then used in Monte Carlo simulations, as explained above.

economy outperforms the stock market the gain generated by the hedge portfolio will offset the increase in the contingent liability.

One might suggest that instead of providing insurance itself the government could promote private provision of GDP-linked minimum return guarantees by issuing GDP-linked bonds. In this way the government would significantly reduce hedging costs and make it easier for the guarantors to transfer the risk of the guarantee to individuals or institutions that are willing to bear it. But the cost estimates in Section 8 already assume that nominal GDP is a traded asset, which means that even if new markets for securities with coupons tied to nominal GDP were to develop, the estimated fair values of GDP-linked minimum return guarantees would not be likely to diminish, unless the new debt instruments significantly increased the correlation between asset returns and GDP growth or reduced their volatilities.

In addition to stimulating the supply of GDP-linked guarantees, such securities would enable pension funds to offer investment lines benchmarked to nominal GDP.<sup>39</sup> However, unlike the publicly guaranteed minimum return, a portfolio tracking nominal GDP would not allow workers to benefit when the financial markets are performing well. An alternative solution could be to require individuals to invest a certain share of their portfolio in GDP-linked bonds, which would undoubtedly be simpler to administer than a publicly guaranteed minimum return. However, the lower the compulsory share of nominal GDP bonds in the portfolio, the less effective the cover against extreme investment risks would be. Conversely, the higher the compulsory portfolio share of nominal GDP bonds, the lower the benefits of having individual funded schemes.

The basic rationale for direct governmental provision of the guarantee is to guard against extreme financial market risks that are not otherwise insurable. In times of systemic shocks, private sector forces are not likely to be able to provide affordable insurance against massive investment risks, so government intervention to protect prospective retirees becomes necessary.

It could be argued that, in such extraordinary circumstances, government can always step in to rescue retirees, especially when participation is not voluntary but compulsory. Public bailouts are indeed an option, provided they are limited to workers who are close to retirement and so will be unable to enjoy the eventual recovery in financial asset prices in the longer term. But rescues are necessarily an extreme remedy, because they are costly for taxpayers and may encourage future opportunistic behaviour by pension funds. For taxpayers a public guarantee scheme funded by risk-based premiums would be more efficient than ex-post bailouts, as the guarantee fund would distribute the burden evenly over time and different cohorts of workers.

Another major factor hampering private provision of such minimum return guarantees is counterparty risk over long-term horizons. Any retirement-related guarantee entails a very long-term commitment. Under such arrangements, DC plan members may well lose confidence in the insurer's ability to stay in the market long enough to fulfill its contractual obligations. The relative inability of market arrangements to offer credible long-term commitment – "market fragility" as Cooper

<sup>&</sup>lt;sup>39</sup> The case for the U.S. government to issue a new security with a coupon tied to current-dollar GDP has now been made by Kamstra and Shiller (2009), who call the new security a win-win solution, as it would benefit both the issuer (the government) and the investors.

and Ross (1999) called it – requires government to step in. Limits to what market forces may achieve also come from the demand side, due to moral hazard problems for the insured. Private financial intermediaries may not be confident about their ability to monitor the behavior of the insured over long-term horizons and may refrain from providing long-term guarantees. A government-backed insurance scheme might manage to reduce moral hazard through regulation and supervision.

The very long-term horizon of a public guarantee fund has other advantages. As is explained in Sections 5 and 7, it allows exploitation of the tendency of asset prices to grow at least as fast as GDP in the long run. And together with government backing, the long-term horizon permits the fund to finance its net outlays through long-term bond issues at advantageous prices and to weather financial market liquidity crises.

One more argument for direct government involvement is uncertainty over the probability distribution of future financial market returns. As Shiller (2006) stresses, the 21st century may differ fundamentally from the 20th. A public guarantee fund seems to be better suited to shoulder such uncertainty over long-term accumulation periods.

Finally, it may also be argued that public guarantee fund's propensity to take risks could be significantly higher than that of the average financial market participant, as it is likely to enjoy better risk-sharing opportunities. This in turn implies that the guarantee fund might set lower risk premiums than private intermediaries.<sup>40</sup>

### **10.** Conclusions

The blow to pension fund assets from the recent financial crisis has underscored how severely members of defined contribution schemes are exposed to financial market tail risks, i.e. to exceptionally large and exceptionally rare drops in financial asset prices that can drastically reduce their accumulated pension value. Market instruments and mutualistic mechanisms may be ineffective and in any case very costly means of protecting returns against these risks. Accordingly, it is worth considering the possible benefits of some form of government-guaranteed minimum return. The public sector's long time horizon and nationwide sphere of action make government best placed to bear the consequences of a collapse in the prices of financial assets. Clearly, such a guarantee system must be structured so as to ensure public financial sustainability.

In the proposal put forward in this article, the government would guarantee a minimum return to defined contribution pension scheme members. The guarantee would be compulsory but would offer several options on the level of the guaranteed return, ranging from nil (guaranteeing the restitution of contributions in nominal terms) to a return as high as the economy's nominal growth rate. Such a scheme would ensure a return on workers' contributions consistent with their risk preferences. In particular, if the guaranteed rate were linked to nominal GDP, the scheme would safeguard future pensioners during the accumulation period against inflation, real shocks to individual sectors of the economy, and a fall in living standards compared

<sup>&</sup>lt;sup>40</sup> For an analysis of the role of risk aversion in the pricing of rate-of-return guarantees for pension funds, see Munnell, Golub-Sass, Kopcke and Webb (2009).

with the rest of the population; workers would also be protected against macroeconomic risks that are otherwise hard to diversify and that could weigh down market returns for protracted periods. At the same time, the guarantee scheme would allow workers to continue to benefit when the net returns on their financial investments exceed the minimum return over the accumulation period. Finally, the guarantee would not require any significant alteration in the operation of supplementary defined contribution pension plans, particularly as regards the retirement saving vehicles of private intermediaries, the availability of different investment lines during the accumulation period, or methods of benefit disbursement.

For the insurance scheme to be financially sustainable, workers must pay a risk-based premium. For it to be economically advantageous as well, the premiums should depend both on the level of the minimum return and on the percentage of risky securities in the portfolio. Government would in any case be able to charge lower premiums than any equivalent private insurance scheme.

Government would manage the swaps through an ad hoc guarantee fund financed out of the reserves generated by workers' premiums and possibly issues of government-backed securities. The guarantee fund would be financially sound, as it would be able to fund its activity at good prices and manage its reserves as a long-term investor.

The existence of a safety net against stock market collapse could of course encourage members of defined contribution pension plans to hold excessively risky portfolios, as in the case of workers close to retirement who are dissatisfied with their returns. This risk of opportunistic behaviour can be averted, say by placing a limit on the proportion of risky securities in a portfolio as the date of pension entitlement approaches. Similarly, the investment choices of asset managers, as well as the governance of the public guarantee fund, should be adequately regulated. In any event, a government-guaranteed minimum return on pension funds would give workers additional incentive to join defined contribution plans and to increase their contribution rate.

## Appendix A: The data used in the estimates of guarantee fund's maximum outlays

The data used in Section 7 to estimate the maximum outlays of the guarantee fund in the United States and Japan are as follows:

- Annual number of workers who retire. It is calculated by dividing the annual number of residents aged 60-64 by five. Source: OECD.Stat Extracts, Labour Force Statistics by sex and age;
- Annual earnings. For the United States: annual mean income (in dollars) of full-time, year-round workers aged 60-64 (before 1993, it is estimated on the basis of the Social Security Actuary's average wage index). Source: U.S. Census Bureau's Current Population Survey (CPS) and Social Security Actuary. For Japan, hourly earnings index in manufacturing and real mean income of working age population (18-65 years). Source: OECD.Stat Extracts, Main Economic Indicators and Social and Welfare Statistics (Income distribution – Inequality);
- *Contribution rate*. Average (employee and employer) contribution rate to 401(k) plans for workers in their sixties in 1999: 12 percent. This is the average

percentage of salary contributed, by employer and employee together, to 401(k) plans in the United States in 1999 for workers in their sixties. Average contribution rates for other age brackets or for plans that do not have employer contributions are much lower (see Table 4 in Holden and VanDerhei, 2001);

- *Gross domestic product in billions of domestic currency.* U.S. Bureau of Economic Analysis and OECD;
- *Total return stock market index*. Shiller (2005) for the United States and Topix Index for Japan (source: Thomson Datastream);
- *Interest rates.* Yield to maturity of 10-year government bond benchmarks (source: Thomson Datastream).

### **Appendix B: The method of estimating the option value**

The method for calculating the value of the put option of a DC plan member is based on risk-neutral valuation (see, for example, sections 11.6 and 16.6 in Hull, 2000) and basically follows Cox and Ross (1976), Merton, Bodie and Marcus (1987), Zurita (1994), Pennacchi (1999), Feldstein and Ranguelova (2001) and Lachance and Mitchell (2003a, 2003b). In order to better model the probability of large changes in stock market indices, share prices are assumed to have jumps superimposed upon a geometric Brownian motion (Merton, 1976).

For comparison with previous studies, the parameter assumptions are similar to those of Lachance and Mitchell (2003a, 2003b). The starting level of annual earnings is set equal to 41,335 US dollars, which corresponds to the Social Security Actuary's average wage index for 2008. The rate of growth of annual earnings is assumed to be 4 percent, while the percentage contribution rate is set equal to 12 percent (see Appendix A). For the risk-free rate, the estimate follows the model in Vasicek (1977). The initial value of the rate, its stationary value, the adjustment factor and the rate volatility are set at 2 percent, 4 percent, 80 percent and 2 percent respectively. The returns on 10-year bonds are estimated with the closed-form model calculated by Vasicek (1977). Share returns in continuous time follow a jump diffusion model in which (assuming risk neutral valuation) the drift is set equal to the risk-free rate adjusted for the average growth rate from the jumps. The discrete time version of the process is:

$$\frac{S(t+\Delta t)}{S(t)} = (1+\rho)^n * exp\left[\left(r(t+\Delta t) - \lambda\rho - \frac{\sigma^2}{2}\right)\Delta t + \sigma\varepsilon\sqrt{\Delta t}\right]$$
(B.1)

where S(t) and r(t) are, respectively, the share price index and the risk-free interest rate at time t,  $\sigma$  is the volatility when a jump does not occur,  $\lambda$  is the rate at which jumps happen,  $\rho$  is the average jump size measured as a proportional increase in the share price index,  $\varepsilon$  is a random sample from a normal distribution with mean zero and standard deviation of 1.0 and n is a random sample from a Poisson distribution with an average number of expected jumps per unit of time equal to  $\lambda$ . The processes  $\varepsilon$  and n are assumed to be independent.

The parameters of the stochastic process for share returns are taken from Di Cesare (2005) and are estimated from the prices of the options on S&P 500 futures contracts listed on the Chicago Mercantile Exchange over the period August 2000-

July 2002. The volatility when a jump does not occur is set equal to 16 percent, the expected number of jumps per year is 1.8 and the average jump size is -12.8 percent (that is, the stock market index falls by almost 13 percent).

In the simulations presented in Table 3 the assumptions are the same as in Table 2 except that share prices follow a geometric Brownian motion without jumps and have a volatility of 20 percent. Nominal GDP is assumed to follow a Brownian geometric motion, with different values for volatility and correlation with share returns.

The final amounts of the notional fund and the pension fund are calculated on the basis of (2) and (3), respectively; the payment generated by the option at the time of retirement is then calculated on the basis of (4). This calculation is made for each of the 10,000 Monte Carlo simulations. The average value of the payments generated by the option in all the simulations is then calculated, and finally its present value assuming risk neutrality is determined. The value obtained is expressed as a percentage of the present value (also calculated assuming risk neutrality) of annual contributions or assets under management.

### Appendix C: The value of the option with a lump-sum investment

It is much easier to calculate the value of the option if we posit a single investment in the pension fund; on this basis, a closed-form calculation is possible. Let us consider this case under the further simplifying assumption that share prices follow a Brownian geometric motion without jumps. The minimum return guarantee linked to the nominal GDP growth rate, which is described in Section 4, corresponds to an option in which one risky asset (the amount accumulated in the pension fund) is swapped with another (the notional fund). This derivative contract, sometimes known as an exchange option, was studied for the first time by Margrabe (1978). It represents a more general case of the classic option  $\dot{a} \, la$  Black-Scholes-Merton, since the strike price is itself a stochastic variable.<sup>41</sup> Assuming the pension fund member makes a single lump-sum payment (represented by  $A_0$ ) at the beginning of the accumulation period, and taking into account that the initial value of the notional fund ( $G_0$ ) is the same as that of the pension fund, the value of the option at the beginning of the accumulation period ( $f_0$ ) is given by the following formula:

$$f_0 = A_0 (N(d_1) - N(d_2))$$
(C.1)

where

$$d_1 = \frac{1}{2}\sqrt{T}\hat{\sigma} \tag{C.2}$$

$$d_2 = d_1 - \sqrt{T}\hat{\sigma} = -\frac{1}{2}\sqrt{T}\hat{\sigma}$$
(C.3)

$$\hat{\sigma} = \sqrt{\sigma_A^2 + \sigma_G^2 - 2\rho_{AG}\sigma_A\sigma_G}$$
(C.4)

and where N(x) is the probability of a standard normal random variable being smaller than x, while  $\hat{\sigma}$ ,  $\sigma_A^2$ ,  $\sigma_G^2$  and  $\rho_{AG}$  are, respectively, the overall volatility of the

<sup>&</sup>lt;sup>41</sup> See also Hull (2000), Chapter 18.

payments generated by the option, the volatility of the pension fund, the volatility of the notional fund, and the correlation between the two stochastic variables.

In equations (C.1-C.4) the price of the option is evidently a function of the overall volatility  $\hat{\sigma}$ , and therefore of the volatility of the two underlying stochastic processes (the pension fund portfolio and nominal GDP) and their correlation. It also obviously reflects the initial value of the insured capital and the duration of the contract.

Given (C.4), the effect of the correlation  $\rho_{AG}$  on total volatility  $\hat{\sigma}$  is monotonic and negative. In fact, it depends on the following condition:

$$\frac{\partial \hat{\sigma}}{\partial \rho_{AG}} \le 0 \iff -2\sigma_A \sigma_G \le 0$$

which always holds. The effect of the correlation on  $\hat{\sigma}$  is annulled if the exercise price of the option is not stochastic ( $\sigma_G = 0$ ), which is the standard Black-Scholes-Merton case.

On the other hand, the effect of the volatility of the strike price ( $\sigma_G$ ) on total volatility  $\hat{\sigma}$  is not monotonic. It depends on the following condition:

$$\frac{\partial \hat{\sigma}}{\partial \sigma_{G}} \ge 0 \iff +2\sigma_{G} - 2\rho_{AG}\sigma_{A} \ge 0 \iff \rho_{AG} \le \min\left(1, \frac{\sigma_{G}}{\sigma_{A}}\right) \tag{C.5}$$

where in (C.5) it is also taken into account that the correlation cannot be greater than 1. Since in the simulations  $\sigma_A \cong 0.105$ , (C.5) implies that

$$\frac{\partial \hat{\sigma}}{\partial \sigma_{G}} \ge 0 \iff \rho_{AG} \le \min(1, 9.52 \, \sigma_{G})$$

(where the last inequality holds with a degree of approximation). Finally, in the extreme case of perfect positive correlation between the value of the portfolio and the option strike price ( $\rho_{AG} = 1$ ), total volatility  $\hat{\sigma}$  is equal to:

$$\hat{\sigma} = \left| \sigma_A - \sigma_G \right| \tag{C.6}$$

and the price of the option is led by the differential between the two volatilities. The value of the exchange option is annulled if two stochastic processes with perfect positive correlation also have the same volatility.

### References

- Antolín, P. and H. Blommestein (2007), "Governments and the Market for Longevity-Indexed Bonds", in *OECD Working papers on insurance and private pensions*, 4. Also published in *OECD Financial Market Trends*, 92.
- Antolín, P. and F. Stewart (2009), "Private Pensions and Policy Responses to the Financial and Economic Crisis", *OECD Working Papers on Insurance and Private Pensions*, 36.
- Blake, D. (2000), "Does it Matter What Type of Pension Scheme You Have?, *Economic Journal*, 110, F46-F81.

- Blake, D., A. Cairns and K. Dowd (2008), "Turning Pension Plans into Pension Planes: What Investment Strategy Designers of Defined Contribution Pension Plans Can Learn from Commercial Aircraft Designers", *Pensions Institute Discussion Paper*, PI-0806.
- Bodie, Z. (1990), "Pensions as Retirement Income Insurance", *Journal of Economic Literature*, 28, 28-49.
- Bodie, Z. (1995), "On the Risks of Stocks in the Long Run", *Financial Analysts Journal*, 51, 18-22.
- Bodie, Z. (2008), "Pension Guarantees, Capital Adequacy and International Risk Sharing", in D. Broeders, S. Eijffinger, A. Houben, *Frontiers in Pension Finance*, Cheltenham, UK: Edward Elgar Publishing.
- Boeri, T., L. Bovenberg, B. Coeuré and A. Roberts (2006), "Dealing with the New Giants: Rethinking the Role of Pension Funds", *Geneva Reports on the World Economy*, 8, Oxford, ICMB-CEPR: Oxford University Press.
- Bonenkamp, J. P. M., M. E. A. J. van de Ven and E. W. M. T. Westerhout (2007), "Macroeconomic Aspects of Intergenerational Solidarity", in O. W. Steenbeek and S. G. van der Lecq (eds.), *Costs and Benefits of Collective Pension Systems*, Berlin: Springer.
- Bütler, M. (2009), "Have social security reforms shifted too much risk to individuals? The financial crisis suggests they might have", *VOX*, website, 13 February 2009, http://www.voxeu.org/index.php?q=node/3059.
- Campbell, J. and M. Feldstein (2001), "Introduction", in J. Campbell and M. Feldstein (eds), *Risk aspects of investment based social security reform*, Chicago: The University of Chicago Press.
- Cesari, R., G. Grande and F. Panetta (2008), "Supplementary Pension Schemes in Italy: Features, Development and Opportunities for Workers", *Giornale degli economisti e annali di economia*, 67, 21-73.
- Chance, D. M. (2003), "Swaptions and Options", Journal of Risk, 5, 67-90.
- Cochrane, J. H. and J. Saá Requejo (2000), "Beyond Arbitrage: Good-Deal Asset Price Bounds in Incomplete Markets", *Journal of Political Economy*, 108, 79-119.
- Coile C., P. B. Levine (2009), "The Market Crash and Mass Layoffs: How the Current Economic Crisis May Affect Retirement", *NBER Working Paper*, 15395.
- Cooper, R. and T. W. Ross (1999), "Public and Private Guarantee Funds with Market Fragility", *The Journal of Risk and Insurance*, 66, 163-184.
- Copeland, C. (2008), "Ownership of Individual Retirement Accounts (IRAs) and 401(k)-Type Plans", *EBRI Notes*, 29, 5.
- Cox, J. C. and S. A. Ross (1976), "The Valuation of Options for Alternative Stochastic Processes", *Journal of Financial Economics*, 3, 145-166.
- Di Cesare, A. (2005), "Estimating Expectations of Shocks Using Option Prices", *The ICFAI Journal of Derivatives Markets*, 2, 42-53.
- Diamond, P. (1977), "A Framework for Social Security Analysis", *Journal of Public Economics*, 8, 275-298.
- Dimson, E., P. Marsh and M. Staunton (2002), *Triumph of the Optimists: 101 Years of Global Investment Returns*, Princeton: Princeton University Press.
- Feldstein, M. and E. Ranguelova (2001), "Accumulated Pension Collars: A Market Approach to Reducing the Risk of Investment-Based Social Security Reform", in

Poterba J., *Tax Policy and the Economy*, 15, 149-166, Cambridge: NBER and MIT Press Journals.

- Fornero, E. (1999), L'economia dei fondi pensione: potenzialità e limiti della previdenza privata in Italia, Bologna: Il Mulino.
- Fornero, E. and E. Luciano (2004), *Developing an Annuity Market in Europe*, Edward Elgar Publishing, Cheltenham.
- Group of Ten (2005), "Ageing and Pension System Reform: Implications for Financial Markets and Economic Policies", Report prepared at the request of the Group of Ten by a group of experts chaired by I. Visco, OECD Financial Market Trends, 89 (Supplement 1). Available at http://www.bis.org/publ/gten09.htm.
- Guazzarotti, G. and P. Tommasino (2008), "The Annuity Market in an Evolving Pension System: Lessons from Italy", *CeRP Working Paper*, 77/08.
- Gustman A. L., T. L. Steinmeier and N. Tabatabai (2009), "What the Stock Market Decline Means for the Financial Security and Retirement Choices of the Near-Retirement Population", *NBER Working Paper*, 15435.
- Holden S. and J. VanDerhei (2001), "Contribution Behavior of 401(k) Plan Participants", *EBRI Issue Brief*, 238.
- Hull, J. C. (2000), *Options, futures and other derivatives*, 4<sup>th</sup> ed., Upper Saddle River, NJ: Prentice-Hall.
- Kaido, H. and H. White (2009), "Inference on Risk-Neutral Measures for Incomplete Markets", *Journal of Financial Econometrics*, 7, 199-246.
- Kamstra, M. and R. Shiller (2009), "The Case for Trills: Giving the People and their Pension Funds a Stake in the Wealth of the Nation", *Cowles Foundation Discussion Paper*, 1717.
- Kotlikoff, L. J. (2008), "The End of Social Security? Don't Reform It, Replace It", *The Boston Globe*, 24 November 2008 (http://kotlikoff.net/content/end-social-security-dont-reform-it-replace-it).
- Lachance, M.-E. and O. S. Mitchell (2003a), "Understanding Individual Account Guarantees", in O. S. Mitchell, K. Smetters, *The Pension Challenge: Risk Transfers and Retirement Income Security*, Oxford: Oxford University Press.
- Lachance, M.-E. and O. S. Mitchell (2003b), "Guaranteeing Individual Accounts", *American Economic Review*, 93, 257-260.
- Lusardi, A. and O. S. Mitchell (2007), Baby Boomer Retirement Security: The Roles of Planning, Financial Literacy, and Housing Wealth, *Journal of Monetary Economics*, 54, 205-224.
- Margrabe, W. (1978), "The Value of an Option to Exchange One Asset for Another", *Journal of Finance*, 33, 177-186.
- Merton, R. C. (1976), "Option Pricing When Underlying Stock Returns Are Discontinuous", *Journal of Financial Intermediation*, 3, 125-44.
- Merton, R. C., Z. Bodie and A. J. Marcus (1987), "Pension Plan Integration as Insurance against Social Security Risk", in Z. Bodie, J. B. Shoven and D. A. Wise, *Issues in Pension Economics*, Chicago: Chicago University Press.
- Modigliani, F. and A. Muralidhar (2004), *Rethinking Pension Reform*, Cambridge: Cambridge University Press.
- Modigliani, F., A. Muralidhar and M. Ceprini (2000), "A Solution to the Social Security Crisis", mimeo, Sloan School of Management.

- Munnell A. H., A. Golub-Sass, R. W. Kopcke and A. Webb (2009), "What Does It Cost To Guarantee Returns?", *Center for Retirement Research at Boston College Issue in Brief*, 8-16.
- OECD (2005), Improving Financial Literacy: Analysis of Issues and Policies, November 2005, Paris, OECD.
- Orszag, P. R. and J. E. Stiglitz (2001), "Rethinking Pension Reform: Ten Myths about Social Security Systems", in R. Holzmann and J. E. Stiglitz (2001), *New Ideas About Old Age Security*, Washington, DC: The World Bank.
- Pennacchi, G. G. (1999), "The Value of Guarantees on Pension Fund Returns", *The Journal of Risk and Insurance*, 66, 219-237.
- Pennacchi, G. G. (2002), "Pension Guarantees: A Methodology for Assessing Fiscal Risk", in H. Polackova Brixi and A. Schick, *Government at Risk*, Washington, DC: The World Bank.
- Pensions Commission (2005), "A New Pension Settlement for the Twenty-First Century – The Second Report of the Pensions Commission", mimeo (www.pensionscommission.org.uk).
- Shiller, R. (1995), "Aggregate Income Risks and Hedging Mechanisms", *The Quarterly Review of Economics and Finance*, 35, 119-152.
- Shiller, R. (2005), *Irrational Exuberance*, 2<sup>nd</sup> edition, Princeton: Princeton University Press. Updated data available at www.econ.yale.edu/~shiller/data.htm.
- Shiller, R. (2006), "Life-Cycle Personal Accounts Proposal for Social Security: An Evaluation of President Bush's Proposal", *Journal of Policy Modeling*, 28, 427-444.
- Steenbeek, O. W. and S. G. van der Lecq (2007), "Summary and Conclusions", in O.W. Steenbeek and S. G. van der Lecq (eds.), *Costs and Benefits of Collective Pension Systems*, Berlin: Springer.
- Turner, J. A. and D. M. Rajnes (2001), "Rate of Return Guarantees for Mandatory Defined Contribution Plans", *International Social Security Review*, 54, 49-66.
- Turner, J. A. and D. M. Rajnes (2003), "Return Guarantees in Voluntary Defined Contribution Plans", in O. S. Mitchell, K. Smetters, *The Pension Challenge: Risk Transfers and Retirement Income Security*, Oxford: Oxford University Press.
- Turner, J. A. and D. M. Rajnes (2009), "Pension Rate of Return Guarantees in a Market Meltdown", mimeo, Pension Policy Center and Social Security Administration, paper presented at the CeRP 10th Anniversary Conference (http://cerp.unito.it/index.php/it/eventi/conferenze/629-conferenza-annuale-cerp).
- VanDerhei, J. (2009), "The Impact of the Recent Financial Crisis on 401(k) Account Balances", *EBRI Issue Brief*, 326.
- Vasicek, O. (1977), "An Equilibrium Characterisation of the Term Structure", *Journal* of Financial Economics, 5, 177-188.
- Visco, I. (2007), "Longevity Risk and Financial Markets", in M. Balling, E. Gnan and F. Lierman (eds), *Money, Finance and Demography: The Consequences of Ageing*, SUERF Colloquium Volume 2006, Vienna: SUERF.
- Visco, I. (2009), "Retirement Saving and the Payout Phase: How To Get There and How To Get the Most Out of It", *OECD Financial Market Trends*, 96.

- Yermo, J. (2008), "Governance and Investment of Public Pension Reserve Funds in Selected OECD Countries", in *OECD Working papers on insurance and private pensions*, 15.
- Zurita, S. (1994), "Minimum Pension Insurance in the Chilean Pension System", *Revista de Análisis Económico*, 9 (1), 105-126.

### RECENTLY PUBLISHED "TEMI" (\*)

- N. 737 *The pro-competitive effect of imports from China: an analysis of firm-level price data*, by Matteo Bugamelli, Silvia Fabiani and Enrico Sette (January 2010).
- N. 738 *External trade and monetary policy in a currency area*, by Martina Cecioni (January 2010).
- N. 739 The use of survey weights in regression analysis, by Ivan Faiella (January 2010).
- N. 740 *Credit and banking in a DSGE model of the euro area*, by Andrea Gerali, Stefano Neri, Luca Sessa and Federico Maria Signoretti (January 2010).
- N. 741 Why do (or did?) banks securitize their loans? Evidence from Italy, by Massimiliano Affinito and Edoardo Tagliaferri (January 2010).
- N. 742 *Outsourcing versus integration at home or abroad*, by Stefano Federico (February 2010).
- N. 743 *The effect of the Uruguay round on the intensive and extensive margins of trade*, by Ines Buono and Guy Lalanne (February 2010).
- N. 744 Trade, technical progress and the environment: the role of a unilateral green tax on consumption, by Daniela Marconi (February 2010).
- N. 745 Too many lawyers? Litigation in Italian civil courts, by Amanda Carmignani and Silvia Giacomelli (February 2010).
- N. 746 On vector autoregressive modeling in space and time, by Valter Di Giacinto (February 2010).
- N. 747 The macroeconomics of fiscal consolidations in a monetary union: the case of Italy, by Lorenzo Forni, Andrea Gerali and Massimiliano Pisani (March 2010).
- N. 748 *How does immigration affect native internal mobility? New evidence from Italy*, by Sauro Mocetti and Carmine Porello (March 2010).
- N. 749 An analysis of the determinants of credit default swap spread changes before and during the subprime financial turmoil, by Antonio Di Cesare and Giovanni Guazzarotti (March 2010).
- N. 750 *Estimating DSGE models with unknown data persistence*, by Gianluca Moretti and Giulio Nicoletti (March 2010).
- N. 751 Down the non-linear road from oil to consumer energy prices: no much asymmetry along the way, by Fabrizio Venditti (March 2010).
- N. 752 Information technology and banking organization, by Sauro Mocetti, Marcello Pagnini and Enrico Sette (March 2010).
- N. 753 *The pricing of government-guaranteed bank bonds*, by Aviram Levy and Andrea Zaghini (March 2010).
- N. 754 Misure di valore aggiunto per le scuole superiori italiane: i problemi esistenti e alcune prime evidenze, by Piero Cipollone, Pasqualino Montanaro and Paolo Sestito (March 2010).
- N. 755 Asset-based measurement of poverty, by Andrea Brandolini, Silvia Magri and Timothy M. Smeeding (March 2010).
- N. 756 Credit supply, flight to quality and evergreening: an analysis of bank-firm relationships after Lehman, by Ugo Albertazzi and Domenico J. Marchetti (April 2010).
- N. 757 A note on rationalizability and restrictions on belief, by Giuseppe Cappelletti (April 2010).
- N. 758 A non-parametric model-based approach to uncertainty and risk analysis of macroeconomic forecasts, by Claudia Miani and Stefano Siviero (April 2010).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- S. SIVIERO and D. TERLIZZESE, *Macroeconomic forecasting: Debunking a few old wives' tales,* Journal of Business Cycle Measurement and Analysis, v. 3, 3, pp. 287-316, **TD No. 395 (February 2001).**
- S. MAGRI, Italian households' debt: The participation to the debt market and the size of the loan, Empirical Economics, v. 33, 3, pp. 401-426, **TD No. 454 (October 2002).**
- L. CASOLARO. and G. GOBBI, *Information technology and productivity changes in the banking industry*, Economic Notes, Vol. 36, 1, pp. 43-76, **TD No. 489 (March 2004).**
- G. FERRERO, *Monetary policy, learning and the speed of convergence,* Journal of Economic Dynamics and Control, v. 31, 9, pp. 3006-3041, **TD No. 499 (June 2004).**
- M. PAIELLA, Does wealth affect consumption? Evidence for Italy, Journal of Macroeconomics, Vol. 29, 1, pp. 189-205, TD No. 510 (July 2004).
- F. LIPPI. and S. NERI, *Information variables for monetary policy in a small structural model of the euro area*, Journal of Monetary Economics, Vol. 54, 4, pp. 1256-1270, **TD No. 511 (July 2004).**
- A. ANZUINI and A. LEVY, *Monetary policy shocks in the new EU members: A VAR approach*, Applied Economics, Vol. 39, 9, pp. 1147-1161, **TD No. 514 (July 2004).**
- D. JR. MARCHETTI and F. Nucci, *Pricing behavior and the response of hours to productivity shocks*, Journal of Money Credit and Banking, v. 39, 7, pp. 1587-1611, **TD No. 524 (December 2004).**
- R. BRONZINI, *FDI Inflows, agglomeration and host country firms' size: Evidence from Italy*, Regional Studies, Vol. 41, 7, pp. 963-978, **TD No. 526 (December 2004).**
- L. MONTEFORTE, Aggregation bias in macro models: Does it matter for the euro area?, Economic Modelling, 24, pp. 236-261, **TD No. 534 (December 2004).**
- A. NOBILI, Assessing the predictive power of financial spreads in the euro area: does parameters instability matter?, Empirical Economics, Vol. 31, 1, pp. 177-195, **TD No. 544 (February 2005).**
- A. DALMAZZO and G. DE BLASIO, *Production and consumption externalities of human capital: An empirical study for Italy*, Journal of Population Economics, Vol. 20, 2, pp. 359-382, **TD No. 554 (June 2005).**
- M. BUGAMELLI and R. TEDESCHI, Le strategie di prezzo delle imprese esportatrici italiane, Politica Economica, v. 23, 3, pp. 321-350, TD No. 563 (November 2005).
- L. GAMBACORTA and S. IANNOTTI, Are there asymmetries in the response of bank interest rates to monetary shocks?, Applied Economics, v. 39, 19, pp. 2503-2517, TD No. 566 (November 2005).
- P. ANGELINI and F. LIPPI, *Did prices really soar after the euro cash changeover? Evidence from ATM withdrawals*, International Journal of Central Banking, Vol. 3, 4, pp. 1-22, **TD No. 581 (March 2006).**
- A. LOCARNO, Imperfect knowledge, adaptive learning and the bias against activist monetary policies, International Journal of Central Banking, v. 3, 3, pp. 47-85, **TD No. 590 (May 2006).**
- F. LOTTI and J. MARCUCCI, *Revisiting the empirical evidence on firms' money demand*, Journal of Economics and Business, Vol. 59, 1, pp. 51-73, **TD No. 595** (May 2006).
- P. CIPOLLONE and A. ROSOLIA, *Social interactions in high school: Lessons from an earthquake*, American Economic Review, Vol. 97, 3, pp. 948-965, **TD No. 596 (September 2006).**
- L. DEDOLA and S. NERI, *What does a technology shock do? A VAR analysis with model-based sign restrictions*, Journal of Monetary Economics, Vol. 54, 2, pp. 512-549, **TD No. 607 (December 2006).**
- F. VERGARA CAFFARELLI, *Merge and compete: strategic incentives for vertical integration*, Rivista di politica economica, v. 97, 9-10, serie 3, pp. 203-243, **TD No. 608 (December 2006).**
- A. BRANDOLINI, Measurement of income distribution in supranational entities: The case of the European Union, in S. P. Jenkins e J. Micklewright (eds.), Inequality and Poverty Re-examined, Oxford, Oxford University Press, TD No. 623 (April 2007).
- M. PAIELLA, The foregone gains of incomplete portfolios, Review of Financial Studies, Vol. 20, 5, pp. 1623-1646, TD No. 625 (April 2007).
- K. BEHRENS, A. R. LAMORGESE, G.I.P. OTTAVIANO and T. TABUCHI, Changes in transport and non transport costs: local vs. global impacts in a spatial network, Regional Science and Urban Economics, Vol. 37, 6, pp. 625-648, TD No. 628 (April 2007).

- M. BUGAMELLI, Prezzi delle esportazioni, qualità dei prodotti e caratteristiche di impresa: analisi su un campione di imprese italiane, v. 34, 3, pp. 71-103, Economia e Politica Industriale, TD No. 634 (June 2007).
- G. ASCARI and T. ROPELE, *Optimal monetary policy under low trend inflation*, Journal of Monetary Economics, v. 54, 8, pp. 2568-2583, **TD No. 647** (November 2007).
- R. GIORDANO, S. MOMIGLIANO, S. NERI and R. PEROTTI, *The Effects of Fiscal Policy in Italy: Evidence from a VAR Model*, European Journal of Political Economy, Vol. 23, 3, pp. 707-733, **TD No. 656** (January 2008).
- B. ROFFIA and A. ZAGHINI, *Excess money growth and inflation dynamics*, International Finance, v. 10, 3, pp. 241-280, **TD No. 657 (January 2008).**
- G. BARBIERI, P. CIPOLLONE and P. SESTITO, Labour market for teachers: demographic characteristics and allocative mechanisms, Giornale degli economisti e annali di economia, v. 66, 3, pp. 335-373, TD No. 672 (June 2008).
- E. BREDA, R. CAPPARIELLO and R. ZIZZA, Vertical specialisation in Europe: evidence from the import content of exports, Rivista di politica economica, v. 97, 3, pp. 189, **TD No. 682 (August 2008).**

- P. ANGELINI, *Liquidity and announcement effects in the euro area*, Giornale degli Economisti e Annali di Economia, v. 67, 1, pp. 1-20, **TD No. 451 (October 2002).**
- P. ANGELINI, P. DEL GIOVANE, S. SIVIERO and D. TERLIZZESE, Monetary policy in a monetary union: What role for regional information?, International Journal of Central Banking, v. 4, 3, pp. 1-28, TD No. 457 (December 2002).
- F. SCHIVARDI and R. TORRINI, *Identifying the effects of firing restrictions through size-contingent Differences in regulation*, Labour Economics, v. 15, 3, pp. 482-511, **TD No. 504 (June 2004).**
- L. GUISO and M. PAIELLA,, *Risk aversion, wealth and background risk*, Journal of the European Economic Association, v. 6, 6, pp. 1109-1150, **TD No. 483 (September 2003).**
- C. BIANCOTTI, G. D'ALESSIO and A. NERI, *Measurement errors in the Bank of Italy's survey of household income and wealth*, Review of Income and Wealth, v. 54, 3, pp. 466-493, **TD No. 520 (October 2004).**
- S. MOMIGLIANO, J. HENRY and P. HERNÁNDEZ DE COS, The impact of government budget on prices: Evidence from macroeconometric models, Journal of Policy Modelling, v. 30, 1, pp. 123-143 TD No. 523 (October 2004).
- L. GAMBACORTA, *How do banks set interest rates?*, European Economic Review, v. 52, 5, pp. 792-819, **TD No. 542 (February 2005).**
- P. ANGELINI and A. GENERALE, *On the evolution of firm size distributions*, American Economic Review, v. 98, 1, pp. 426-438, **TD No. 549 (June 2005).**
- R. FELICI and M. PAGNINI, *Distance, bank heterogeneity and entry in local banking markets*, The Journal of Industrial Economics, v. 56, 3, pp. 500-534, No. 557 (June 2005).
- S. DI ADDARIO and E. PATACCHINI, *Wages and the city. Evidence from Italy*, Labour Economics, v.15, 5, pp. 1040-1061, **TD No. 570 (January 2006).**
- S. SCALIA, *Is foreign exchange intervention effective?*, Journal of International Money and Finance, v. 27, 4, pp. 529-546, **TD No. 579 (February 2006).**
- M. PERICOLI and M. TABOGA, Canonical term-structure models with observable factors and the dynamics of bond risk premia, Journal of Money, Credit and Banking, v. 40, 7, pp. 1471-88, TD No. 580 (February 2006).
- E. VIVIANO, *Entry regulations and labour market outcomes. Evidence from the Italian retail trade sector*, Labour Economics, v. 15, 6, pp. 1200-1222, **TD No. 594 (May 2006).**
- S. FEDERICO and G. A. MINERVA, Outward FDI and local employment growth in Italy, Review of World Economics, Journal of Money, Credit and Banking, v. 144, 2, pp. 295-324, TD No. 613 (February 2007).
- F. BUSETTI and A. HARVEY, *Testing for trend*, Econometric Theory, v. 24, 1, pp. 72-87, **TD No. 614** (February 2007).

- V. CESTARI, P. DEL GIOVANE and C. ROSSI-ARNAUD, Memory for prices and the Euro cash changeover: an analysis for cinema prices in Italy, In P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, TD No. 619 (February 2007).
- B. H. HALL, F. LOTTI and J. MAIRESSE, Employment, innovation and productivity: evidence from Italian manufacturing microdata, Industrial and Corporate Change, v. 17, 4, pp. 813-839, TD No. 622 (April 2007).
- J. SOUSA and A. ZAGHINI, *Monetary policy shocks in the Euro Area and global liquidity spillovers,* International Journal of Finance and Economics, v.13, 3, pp. 205-218, **TD No. 629 (June 2007).**
- M. DEL GATTO, GIANMARCO I. P. OTTAVIANO and M. PAGNINI, Openness to trade and industry cost dispersion: Evidence from a panel of Italian firms, Journal of Regional Science, v. 48, 1, pp. 97-129, TD No. 635 (June 2007).
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "inflation perceptions"? A survey-based analysis of Italian consumers, in P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, TD No. 655 (January 2008).
- R. BRONZINI, G. DE BLASIO, G. PELLEGRINI and A. SCOGNAMIGLIO, *La valutazione del credito d'imposta per gli investimenti*, Rivista di politica economica, v. 98, 4, pp. 79-112, **TD No. 661 (April 2008).**
- B. BORTOLOTTI, and P. PINOTTI, *Delayed privatization*, Public Choice, v. 136, 3-4, pp. 331-351, **TD No.** 663 (April 2008).
- R. BONCI and F. COLUMBA, Monetary policy effects: New evidence from the Italian flow of funds, Applied Economics, v. 40, 21, pp. 2803-2818, TD No. 678 (June 2008).
- M. CUCCULELLI, and G. MICUCCI, *Family Succession and firm performance: evidence from Italian family firms*, Journal of Corporate Finance, v. 14, 1, pp. 17-31, **TD No. 680 (June 2008).**
- A. SILVESTRINI and D. VEREDAS, *Temporal aggregation of univariate and multivariate time series models: a survey*, Journal of Economic Surveys, v. 22, 3, pp. 458-497, **TD No. 685 (August 2008).**

- F. PANETTA, F. SCHIVARDI and M. SHUM, Do mergers improve information? Evidence from the loan market, Journal of Money, Credit, and Banking, v. 41, 4, pp. 673-709, TD No. 521 (October 2004).
- M. BUGAMELLI and F. PATERNÒ, *Do workers' remittances reduce the probability of current account reversals?*, World Development, v. 37, 12, pp. 1821-1838, **TD No. 573 (January 2006).**
- P. PAGANO and M. PISANI, *Risk-adjusted forecasts of oil prices*, The B.E. Journal of Macroeconomics, v. 9, 1, Article 24, **TD No. 585 (March 2006).**
- M. PERICOLI and M. SBRACIA, The CAPM and the risk appetite index: theoretical differences, empirical similarities, and implementation problems, International Finance, v. 12, 2, pp. 123-150, TD No. 586 (March 2006).
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and the business cycle*, Journal of Financial Stability, v. 5, 4, pp. 393-409, **TD No. 601 (September 2006).**
- S. MAGRI, *The financing of small innovative firms: the Italian case*, Economics of Innovation and New Technology, v. 18, 2, pp. 181-204, **TD No. 640 (September 2007).**
- S. MAGRI, *The financing of small entrepreneurs in Italy*, Annals of Finance, v. 5, 3-4, pp. 397-419, **TD** No. 640 (September 2007).
- V. DI GIACINTO and G. MICUCCI, The producer service sector in Italy: long-term growth and its local determinants, Spatial Economic Analysis, Vol. 4, No. 4, pp. 391-425, TD No. 643 (September 2007).
- F. LORENZO, L. MONTEFORTE and L. SESSA, *The general equilibrium effects of fiscal policy: estimates for the euro area*, Journal of Public Economics, v. 93, 3-4, pp. 559-585, **TD No. 652 (November 2007).**
- R. GOLINELLI and S. MOMIGLIANO, *The Cyclical Reaction of Fiscal Policies in the Euro Area. A Critical Survey of Empirical Research*, Fiscal Studies, v. 30, 1, pp. 39-72, **TD No. 654 (January 2008).**
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "Inflation Perceptions"? A survey-based analysis of Italian consumers, Giornale degli Economisti e Annali di Economia, v. 68, 1, pp. 25-52, TD No. 655 (January 2008).

- F. MACCHERONI, M. MARINACCI, A. RUSTICHINI and M. TABOGA, *Portfolio selection with monotone meanvariance preferences*, Mathematical Finance, v. 19, 3, pp. 487-521, **TD No. 664 (April 2008).**
- M. AFFINITO and M. PIAZZA, What are borders made of? An analysis of barriers to European banking integration, in P. Alessandrini, M. Fratianni and A. Zazzaro (eds.): The Changing Geography of Banking and Finance, Dordrecht Heidelberg London New York, Springer, TD No. 666 (April 2008).
- G. FERRERO and A. NOBILI, *Futures contract rates as monetary policy forecasts*, International Journal of Central Banking, v. 5, 2, pp. 109-145, **TD No. 681 (June 2008).**
- L. ARCIERO, C. BIANCOTTI, L. D'AURIZIO and C. IMPENNA, *Exploring agent-based methods for the analysis* of payment systems: A crisis model for StarLogo TNG, Journal of Artificial Societies and Social Simulation, v. 12, 1, **TD No. 686 (August 2008).**
- A. CALZA and A. ZAGHINI, *Nonlinearities in the dynamics of the euro area demand for M1*, Macroeconomic Dynamics, v. 13, 1, pp. 1-19, **TD No. 690 (September 2008).**
- L. FRANCESCO and A. SECCHI, *Technological change and the households' demand for currency*, Journal of Monetary Economics, v. 56, 2, pp. 222-230, **TD No. 697 (December 2008).**
- G. ASCARI and T. ROPELE, *Trend inflation, taylor principle, and indeterminacy*, Journal of Money, Credit and Banking, v. 41, 8, pp. 1557-1584, **TD No. 708** (May 2007).
- S. COLAROSSI and A. ZAGHINI, *Gradualism, transparency and the improved operational framework: a look at overnight volatility transmission*, International Finance, v. 12, 2, pp. 151-170, **TD No. 710 (May 2009).**
- M. BUGAMELLI, F. SCHIVARDI and R. ZIZZA, *The euro and firm restructuring*, in A. Alesina e F. Giavazzi (eds): Europe and the Euro, Chicago, University of Chicago Press, **TD No. 716 (June 2009).**
- B. HALL, F. LOTTI and J. MAIRESSE, *Innovation and productivity in SMEs: empirical evidence for Italy*, Small Business Economics, v. 33, 1, pp. 13-33, **TD No. 718 (June 2009).**

- S. MAGRI, *Debt maturity choice of nonpublic Italian firms*, Journal of Money, Credit, and Banking, v.42, 2-3, pp. 443-463, **TD No. 574 (January 2006).**
- S. NERI and A. NOBILI, *The transmission of US monetary policy to the euro area*, International Finance, v. 13, 1, pp. 55-78, **TD No. 606 (December 2006).**
- M. IACOVIELLO and S. NERI, *Housing market spillovers: evidence from an estimated DSGE model,* American Economic Journal: Macroeconomics, v. 2, 2, pp. 125-164, **TD No. 659 (January 2008).**
- A. B. ATKINSON and A. BRANDOLINI, *On analysing the world distribution of income*, World Bank Economic Review, v. 24, 1, pp. 1-37, **TD No. 701 (January 2009).**
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, *Asset-based measurement of poverty*, Journal of Policy Analysis and Management, v. 29, 2, pp. 267-284, **TD No. 755** (March 2010).

### FORTHCOMING

- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, **TD No. 458 (December 2002).**
- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD No.** 578 (February 2006).
- G. DE BLASIO and G. NUZZO, *Historical traditions of civicness and local economic development*, Journal of Regional Science, **TD No. 591 (May 2006).**
- R. BRONZINI and P. PISELLI, *Determinants of long-run regional productivity with geographical spillovers: the role of R&D, human capital and public infrastructure,* Regional Science and Urban Economics, **TD No. 597 (September 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, **TD No. 598 (September 2006).**
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing Excess Liquidity in the Euro Area: The Role of Sectoral Distribution of Money, Applied Economics, **TD No. 627** (April 2007).

- A. CIARLONE, P. PISELLI and G. TREBESCHI, *Emerging Markets' Spreads and Global Financial Conditions*, Journal of International Financial Markets, Institutions & Money, **TD No. 637 (June 2007)**.
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS, *Securitisation and the bank lending channel*, European Economic Review, **TD No. 653 (November 2007).**
- F. BALASSONE, F. MAURA and S. ZOTTERI, *Cyclical asymmetry in fiscal variables in the EU*, Empirica, **TD** No. 671 (June 2008).
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, **TD No. 673 (June 2008).**
- F. D'AMURI, O. GIANMARCO I.P. and P. GIOVANNI, *The labor market impact of immigration on the western* german labor market in the 1990s, European Economic Review, **TD No. 687 (August 2008).**
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, **TD No. 688 (September 2008).**
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector: the case of Italy*, Macroeconomic Dynamics, **TD No. 706 (March 2009).**
- Y. ALTUNBAS, L. GAMBACORTA, and D. MARQUÉS-IBÁÑEZ, *Bank risk and monetary policy*, Journal of Financial Stability, **TD No. 712 (May 2009).**
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Dynamic macroeconomic effects of public capital: evidence from regional Italian data, Giornale degli economisti e annali di economia, TD No. 733 (November 2009).
- F. COLUMBA, L. GAMBACORTA and P. E. MISTRULLI, *Mutual Guarantee institutions and small business finance*, Journal of Financial Stability, **TD No. 735** (November 2009).
- A. GERALI, S. NERI, L. SESSA and F. M. SIGNORETTI, *Credit and banking in a DSGE model of the Euro Area,* Journal of Money, Credit and Banking, **TD No. 740 (January 2010).**
- L. FORNI, A. GERALI and M. PISANI, *The macroeconomics of fiscal consolidations in euro area countries,* Journal of Economic Dynamics and Control, **TD No. 747** (March 2010).